

# Is destiny worth the Distance? On Private Equity in Emerging Markets

Sara Ain Tommar, Serge Darolles, E. Jurzcenko

# ► To cite this version:

Sara Ain Tommar, Serge Darolles, E. Jurzcenko. Is destiny worth the Distance? On Private Equity in Emerging Markets. 3rd International Bordeaux Workshop in Quantitative Finance, Risk, and Decision Theor, Nov 2018, Bordeaux, France. hal-04010167

# HAL Id: hal-04010167 https://hal.science/hal-04010167

Submitted on 1 Mar 2023

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

# Is Destiny Worth the Distance? On Private Equity in Emerging Markets\*

March 2018 (Preliminary Version)

Sara Ain Tommar<sup>†</sup>

Serge Darolles<sup>‡</sup>

Emmanuel Jurczenko§

#### Abstract

We study the performance determinants of private equity investing in emerging markets (EM) compared to developed markets (DM) using a novel dataset. Using a multilevel linear model specification, our results suggest that performance in emerging markets in highly dependent on geographical and cultural proximity. The effect is significantly higher for GPs investing in both markets compared to pure DM- and EM-players respectively. Cross-cultural and geographical effects are enhanced when the GP investment teams are also culturally close using different measures. Our results also show that the realized returns are highly dependent on the investment period, the investment style and the GP's experience on each market.

Keywords: Private Equity investing, Private Equity performance, Emerging Markets

<sup>\*</sup> With support from the Private Equity and Venture Capital Research Initiative, under the aegis of the Europlace Institute of Finance, whom we gratefully thank.

This research is inspired by a suggestion from Zsuzsanna Fluck.

<sup>&</sup>lt;sup>†</sup> Corresponding Author.

Sara Ain Tommar is a PhD candidate in Finance at Université Paris Dauphine–PSL Research University. Place du Maréchal de Lattre de Tassigny, 75016 Paris, France. E-mail : <u>sara.ain-tommar@dauphine.eu</u>

<sup>&</sup>lt;sup>‡</sup> Serge Darolles is a Professor of Finance at Université Paris Dauphine–PSL Research University. Place du Maréchal de Lattre de Tassigny, 75016 Paris, France. E-mail : <u>serge.darolles@dauphine.fr</u>

<sup>&</sup>lt;sup>§</sup> Emmanuel Jurczenko is Associate Dean & Clinical Professor in Finance, and Director of Institute for Hospitality Real Estate & Finance at Ecole Hôtelière de Lausanne. Route de Cojonnex 18/1000 Lausanne 25/ Switzerland. E-mail : Emmanuel.JURCZENKO@ehl.ch

"Much of the growth of venture capital and private equity activity is going to take place in emerging markets".

Josh Lerner, The Future of Private Equity, European Financial Management, Vol. 17, Issue 3, pp. 423-435, 2011.

## 1. Introduction

Recent years have seen steady growth of private equity (PE) investing in emerging markets (EM), attracted by the growing population, the steadily developing middle-class and GDP growth perspectives. In the aftermath of the financial crisis, many of the industry's giants were tempted by the potential of these markets, where financial markets and banking systems are often not sufficiently developed to meet local companies financing needs, and where capital markets offer more exit opportunities with the growing numbers of IPOs. Mediated examples include KKR and Carlyle, who struck multi-million dollar deals in Africa among other emerging markets destinations. Additionally, according to EMPEA's LP survey<sup>2</sup>, 75% of limited partners (LPs) with private equity allocation to emerging markets-focused funds declare still eying investments in these markets.

Vendor databases and the business press note the growing interest in emerging markets private equity, with mitigated results about their prospected outperformance<sup>3</sup>. Research on private equity in emerging markets has long been challenged by the scarcity of data. Most existing studies use country-level data (e.g. Mexico

<sup>&</sup>lt;sup>2</sup> 2017 Limited Partners Survey, EMPEA (Emerging Markets Private Equity Association).

<sup>&</sup>lt;sup>3</sup> See Preqin Special Reports, "Private Equity in Emerging Markets", July 2017 and Forbes, "Private Equity shifting gears in emerging markets", April 22, 2014, among others.

(Charvel, 2012), India (Gohil, 2014), Brazil (Minardi et al., 2014) with mitigated results on private equity performance in those countries. More recent research looks at larger sets of emerging markets economies Johan and Zhang (2016) and Lerner and Baker (2017).

In this paper, we aim to contribute to the growing body of literature on emerging markets in general (Karolyi, 2016) and in private equity in particular; using opportunities offered by new data. We aim to investigate if there are any substantial performance differentials between private equity in developed markets (DM) and private equity in emerging markets (EM), then define the investment determinants and performance drivers of such observed differences.

This research contributes to the existing literature on private equity in general, and on private equity in emerging markets in particular, in three ways. First, we place our focus at the portfolio company by studying performance at the deal level. As pointed out by Braun et al. (2016) and Braun et al. (2017), uncovering underlying investments rather than looking at the funds is a substantial advantage when studying private equity performance.

Second, our study is large scale. Private equity is a dynamic asset class in a naturally growing global investment universe. Comparing realized returns across GPs, countries and portfolio companies is subject to provide insights into how these dynamics are shaped within the structuring of a fund and in the backing channel. Moreover, the gross return as a measure is in our view a good point estimates in a context of increased demand for lower fees and growing direct investments by a private equity LPs<sup>4</sup>.

<sup>4</sup> A Private Equity International article (dated January 25<sup>th</sup>, 2018) and an interesting analysis by Carlyle's David Rubenstein in Reuters' November 17<sup>th</sup>, 2017 article can be found here: <u>https://www.privateequityinternational.com/buyout-firms-unfazed-by-competition-from-lps-on-deals/</u> <u>https://www.reuters.com/article/carlyle-group-loans/lpc-private-equity-to-face-competition-from-investors-carlyle-co-ceo-</u> idUSL8N1NN3LT Third, we contribute to the nascent body of literature on emerging markets given availability of new data and through a new lens. Using a sample of 2733 exits in 35 emerging markets, Johan and Zhang (2016) show that successful private equity exits are conditioned by better business and legal environments. Using a transaction-level TVPI measure, Lerner and Becker (2017) show that there is less dispersion in emerging markets private equity returns compared to developed markets returns. In this paper, we try to join both efforts by uncovering private equity investment determinants and performance outcomes using proximity and direction of fund flows indicators. We argue that even though private markets are more and more global, and even within seemingly homogenous economic groups (EMs or DMs), geographical or cultural differences may have a role in shaping business relationships and in encouraging investment flows from a country to another. Moreover, we argue that beyond countries, individual qualities, such as culturally proximate investment teams for example, may further accentuate this effect. This view further extends the one in Johan and Zhang (2016) by adding up to the conclusions drawn on the effect of institutional quality at the country level on achieving better exits. We examine culture and geography as ex-ante investment determinants from observing realized returns ex-post, controlling for performance differentials factors. We also regard both markets (DMs and EMs) geographically as already set diversified portfolios for investors. As discussed earlier, seemingly homogenous investment universes vary greatly in local cultural dimensions, which might offer diversification benefits combined to geography. Nevertheless, the risks associated with any new investment requires experience and *proximity* from the GPs, which we are able to measure with granularity in the detail of the backing channel: private equity investments are first and foremost investments made in passionate people (VCs) or successful corporations (buyouts), by seasoned investment individuals

(investment professionals). We interact those dimensions and have them in the focus of our lens throughout the study.

We collect data from various databases on private equity fund, investments, GPs and individual managers and augment them with cultural and geographical data. We structure our data from all sources in a way that allows us to cleanly observe the investment channel: LP > GP > Fund > Portfolio Company. We are also able to observe the profiles of the individual managers within the GPs at most investment periods and the geographical locations of each GP, fund and portfolio company. We use a multilevel linear model approach to account for this hierarchy in our data structure and also to separate each level's effect on performance (in a top down approach, starting from the base : portfolio company effects, country effects and GP effects). We group the GP's and portfolio companies' countries geographically as local, nearby or distant depending on whether the GP's and portfolio company's countries are located in the same country, do, or do not share a maritime or land border respectively. We also use textual analysis to process the profiles of investment individuals and cross-interact them with country-level cultural and geographical dimensions to uncover effects of closer (respectively more distant) teams culturally on farther (respectively closer) investments geographically. We also test whether previous geographical concentration of the GP prior to the follow-on investments affects the outcome of those investments, using a Hirschman-Herfindahl measure of geographically allocated capital across countries.

Our results indicate that there are significant cultural and geographical effects that shape the investment directions of private capital flows, alongside previously documented performance drivers in the private equity literature (notably the *money chasing deals hypothesis*, Gompers and Lerner, 2000). Our results are especially true for the most significant economic sub-period on both markets (i.e. investments made after 2000) and show

predictability over time. These results are robust to a correction for possible gaps in deal sequences and to the use of other performance measures (multiple of invested capital and successful exit rates at the fund level).

The rest of the paper is organized as follows: section 2 provides the background of our study and discusses some of the literature that relates to our setting, section 3 describes the methodology of our study, section 4 discusses some of the summary statistics from our sample. Empirical findings are provided is section 5 and section 6 concludes.

# 2. Background

### 2.1. Emerging markets and Private Equity performance

The definition of an emerging market is central to our research question and is relatively problematic given that there is no consensus on what an emerging market is. The term *emerging market* was first mentioned by Dutch economist Antoine W. Van Agtmael in 1981 and was picked up since to loosely designate a country with low income per capita and high expected economic growth. There is no official listing of emerging markets, but investment professionals usually refer to one of the following five listings: the international monetary fund (23 countries), Morgan Stanley Capital International (also 23 countries), the Dow Jones (22 countries), Standard and Poor's (21 countries), and Russel (also 21 countries).

The recent years witnessed growing interest in emerging markets among investors in an attempt to leverage on the growth perspectives of these markets in a highly matured developed economy. However, the risks associated with those markets on the one hand, and the needed adjustment effort from foreign GPs on the other hand<sup>5</sup>, lead to mitigated investment experiences and mixed conclusions among investors on how emerging markets are exactly an investment opportunity<sup>6</sup>. In the literature, issues related to scarcity and quality of data often held back the development of this yet interesting research field, with most studies often restricted to single-country level studies (Charvel, 2012, Gohil, 2014, Minardi et al., 2014 among others).

<sup>&</sup>lt;sup>5</sup> E.g. KKR stepped out of the African market when it failed to source sufficiently sizeable deals, and the Carlyle Group, historically a buyout firm, shifted to growth strategies and minority positions in some of emerging markets most conservative, family-dominated countries.

<sup>&</sup>lt;sup>6</sup> See for example the Financial Times' November 6<sup>th</sup>, 2017 article: *Private Equity turns to Asia's frontier market for growth*, <u>https://www.ft.com/content/8db6c03e-b497-11e7-aa26-bb002965bce8</u> (Visited December 20<sup>th</sup>, 2017)

Recent efforts initiated larger-scale studies (Johan and Zhang, 2016, Lerner and Becker, 2017), offered by new data opportunities. This paper is in the continuum of this nascent literature and aims to take advantage of the granularity of a newly structured dataset to disentangle GP, country and deal characteristics' effects on private equity performance.

### 2.2. Geography, Culture, and Private Equity performance

#### 2.2.1. Investment performance and geographical influences

There is an extensively grown body of literature on distance and investment performance in the asset management literature. Coval and Moskowitz (2001) show that mutual fund managers earn higher returns from nearby investments compared to distant investments. Investors also prefer geographically proximate investments and they overweigh domestic holdings in their portfolios (Coval and Moskowitz, 1999). In equity analysis, Malloy (2005) shows that US equity analysts are better at earning forecasts for nearby than for distant firms. Instances of geographical preferences in a private equity setting are documented in Chen et al. (2010), where the authors show increased geographical concentration of VC investors and VC-backed portfolio companies in three major US cities, and that VC investments outperform where VC firms are in VC centers. The scopes of such studies increased both in terms of geographical focus and country coverage. Using Korean data, Choe et al. (2005) show that local investors outperform foreign investors in trading local stocks. The same conclusion is drawn in a European market: local traders have an edge over foreign traders in Germany (Hau, 2001). Along the lines of the findings of Malloy (2005) in the US market, Bae et al. (2008) show that local analysts issue better earnings forecasts than geographically distant analysts, using data on 32 countries. An interesting side-result is that the observed effect is stronger for emerging markets, where they argue firms are highly opaque and disclose less information. In the hedge fund literature, Teo (2009) documents significant outperformance of hedge funds which are geographically proximate to their investments.

Besides differences induced by geographical effects, the literature has also documented cultural influences on investment performance.

#### 2.2.2. Investment performance and cultural influences

In a general setting, countries have been shown to significantly impact investment and doing business in the seminal papers of Shleifer and Vishny (1997) and La Porta et al. (1997, 2012), depending on factors such legal origin or the level of investor protection. Subsequent literature has shown that these values are shaped by cultural dimensions (Licht, Goldschmidt, and Schwartz, 2005, 2007).

As culture is a complex and multi-dimensional construct, researchers often rely on models and scores developed in sociology to gauge the complexity of cultural values. There instances of usually four cultural models, Hofstede's (2001) dimensions, Schwartz's (1994) cultural model, The World Value Survey and The Global Leadership & Organizational Behavior Effectiveness (GLOBE) study (House et al. 2004).

Instances of the use of language as a driver for cultural values has been used in the asset management literature : traders from outside Germany in non-speaking German cities underperform traders located in Germany and in financial centers such as Frankfurt (Hau, 2005). Teo (2009) also documents that hedge funds

with native speaking managers in the Asian market overperform. We base our use of language at the individual level as a proxy for cultural proximity within the investment teams of the GPs in our private equity setting and rely on Schwartz's cultural dimensions as a more appropriate model for our country level characteristics subset (Ng et al. 2006).

# 3. Methodology

### 3.1. Sample Structure

To assess performance differentials between emerging markets private equity and developed markets private equity, we focus on GPs with investment experience on both markets (treatment group) and study their investment determinants and performance relative to GPs with investment histories solely in either developed markets or emerging markets (control groups). To this end, we start by geographically grouping the GPs and the portfolio companies based on their respective headquarters location. To classify countries as either a developed or emerging market, we source the previously cited EM listings dynamically over time and cross-reference them with the World Bank's Income Groups (i.e. economic regions) to account for the most common criteria of each EM listing. We consider only the World Bank sovereign member countries (189 out of 196 currently).

### [Figure 1 about here]

As we hypothesize about cultural proximity being a driver of private equity performance in emerging markets (and also possibly within developed markets), we give special care to within-country cultural differences and assign GPs and portfolio companies geographically by city when applicable<sup>7</sup> and where the amount of invested capital is significant.

Next, we establish a *deal direction* measure and an *investment direction* measure. The deal direction takes the values DMtoDM, DMtoEM, EMtoDM and EMtoEM following the regional location of the origin of capital (GP) and the destination of capital (portfolio company). Next, we aggregate the Deal Directions by GP to establish an investment direction, which takes the values DMtoDM, DMtoDM&EM, DMtoEM for DM-based GPs, and EMtoDM, EMtoEM for EM-based GPs. Naturally, the Investment Direction depends on observed instances of one or multiple deal directions for each GP.

We note the treatment group DMtoDM&EM (i.e. DM-based GPs with investment history in DM and EM), and the control groups respectively DMtoDM (i.e. DM-based GPs with observable investment history in DM only) and EMtoEM (i.e. EM-based GPs with observable investment history is EM only). Cases where EM-based GPs invest exclusively in developed markets (EMtoDM), or those headquartered in DM with observed investments in EM only (DMtoEM) are excluded them from the analysis because of their lack of economic significance (11 DM-based GPs with less than 10 million dollars invested in EM over the sample period, and 6 EM-based GPs with an aggregated 7 million dollars investments in DM). Figure 1 provides a visual for our sample structure by deal and investment directions.

These groupings allow us to assess whether performance differentials pertain to GP characteristics -in which case significant effects would be noted in the cross-section of GPs between the treatment group and the control group- or to other intrinsic characteristics : the portfolio company, the country and/or region. In the

<sup>&</sup>lt;sup>7</sup> In developed markets, examples include English- vs. French-speaking Canada, or Wallonia and Flanders in Belgium. In emerging markets, an example is India, where Hindi and English are both considered official languages (as stated by the Official Languages Act, 1963 – amended 1987, viewed on the Indian Ministry of Electronics and Information Technology.)

latter case, significant performance differentials would be observed in the cross-section of investments within the treatment group, controlling for GP, country, and company characteristics.

### 3.2. Data and variable definition

Our database contains detailed information on the geography of managers and portfolio companies, as well as financing data and performance metrics. We source performance data from Pitchbook and Preqin on managers, private equity funds and portfolio companies, with more details on the latter from Orbis-BvD, SDC and Zephyr-BvD, along descriptive data from Thomson Financials. The quality and relevance of these datasets to private equity performance studies are discussed in Ain Tommar and Darolles (2017).

With regards to investment determinants, we conjecture whether proximity plays a role in targeting these markets. We use two measures for proximity: geographical proximity and cultural proximity. Geographical proximity is measured using a category variable on whether the GP is local, nearby or distant to/from their investment, following common land or maritime borders shared by the headquarter countries of the GP and the portfolio company. Cultural proximity is proxied in linguistic and reference model terms. Using language, we look for target-investment-country language speakers in the investment teams of the GP at the time of the deal. On the target country side, we collect information on locally spoken languages from the World Fact Book<sup>8</sup>. An interesting feature of these data is that it provides the *lingua franca* and the *commonly understood* languages for each country alongside the official languages<sup>9</sup>. On the GP side, we start by looking for

<sup>&</sup>lt;sup>8</sup> <u>https://www.cia.gov/library/publications/the-world-factbook/fields/2098.html</u>, visited on June 6<sup>th</sup>, 2017.

<sup>&</sup>lt;sup>9</sup> Another interesting feature is that languages come with the history and heritage of countries and are subject to tens if not hundreds of years in order to change, which works against endogeneity among other issues.

investment professionals with the "deal makers" tag in the universe of Capital IQ's People Intelligence Database and we name-match the companies in their employment histories to our list of GPs, accounting for name changes, AKAs, FKAs<sup>10</sup> and M&As between GPs. We also cross-reference the obtained list of GP investment professionals to their available biographies from Pitchbook<sup>11</sup>. We analyze the probability for an investment professional to speak the language of the target country at two level: the name level and the biography level. Using names, we use a language identification API to textually process the names of the investment professionals and assess their *resonance* in order to establish native languages. We also assume that the investment professionals speak the languages of where every company in their employment history is located. Using biographies, we mainly look for the educational background of each investment professional to establish the location of their schools and therefore the language of the country where they are located. the People Intelligence database also gives the employment start dates and end dates. When available, we cross-reference these dates to the deal making window at the portfolio company level to assess how the presence of a language-proficient team member within the GP investment teams can close a possible cultural gap in the deal making process.

We use three measures to assess the intensity of cultural proximity at the GP level. A dummy variable for whether there is a language proficient speaker among the teams of the GP at the time of the deal. A level variable for the number of language proficient speakers in the GP's investment teams, and a concentration variable for a Hirschman-Herfindahl measure of the number of language proficient team members relative to the investment team size. Appendix 1 provides a detailed review of all used variables and table 8 shows

<sup>&</sup>lt;sup>10</sup> « Also Known As » and « Frequently Known As ».

<sup>&</sup>lt;sup>11</sup> Pitchbook lists the investment professionals at the fund level. Given our data structure we are able to identify them to GPs and deals respectively.

the correlations between the GP characteristics, between the country characteristics and between the portfolio company characteristics.

Using a reference cultural model, we base our analysis on Schwartz's (1994, 2004) cultural dimensions and we focus on values which are more likely to reflect institutional quality: embeddedness, hierarchy, mastery and intellectual intelligence. Definitions of these cultural dimensions are given in appendix 1 and their scores are available for 80 countries on the author's webpage<sup>12</sup>. We use the difference of these scores between the country where the GP is located and the country of the portfolio company as independent variables ate the country level.

With regards to performance, we use performance measures both at the fund and portfolio company levels using measures from the previous literature on private equity performance given in table 1.

#### [Table 1 about here]

Given the setting of our study and challenges associated with data availability and quality, we mainly focus on the deal-level PME as the primary measure and provide results using the multiple of invested capital and successful exit rates in the robustness checks in tables 14 and 15. Our results generally hold although with not as much a significant economic amplitude in the exit rate or the multiple of invested capital compared to deal-level PMEs.

<sup>&</sup>lt;sup>12</sup>https://www.researchgate.net/publication/304715744 The 7 Schwartz cultural value orientation scores for 80 countries, visited on February 3<sup>rd</sup>, 2018.

As highlighted before, our estimation of performance is two-fold. We investigate whether private equity returns in emerging markets are substantially different from those in developed markets across different groups of GPs and within the same group of GPs that have investment experience on both markets. To assess whether returns are sensitive to the GP's choice of investing in emerging markets compared to investing in their home countries, we estimate a multilevel linear model of our return measures against GP, country, and portfolio company level characteristics respectively, augmented by other possible performance drivers and investment determinants variables.

#### 3.3. A Multilevel linear model

Motivations for a multilevel linear model empirical specification is simultaneously driven by the structure of the data and the nature of the study. First, our data is multilevel. Portfolio companies are nested in 165 countries which are either developed or emerging and are differently "proximate" to GPs. At the GP level, we have 11,807 GPs for up to 50 years. At the country level, companies are nested in 165 countries and at the portfolio company level, GP firms are invested in 70,696 companies. To separate the within-country and across-country effects of GP-level variables such as EM-investment vs. DM-investment on GP performance, we use the following multilevel linear model specification.

$$y_{ijkt} = \alpha_{jk} + x'_{it} \beta + \varepsilon_{ijkt}$$
$$\alpha_{jk} = \nu_k + w'_j \gamma + u_{jk}$$
$$\nu_k = z'_k \gamma + \eta_k$$

Where  $y_{ijkt}$  is the deal-level performance measure of GP *i* in country *j* and company *k* at time *t*,  $x_{it}$  is a vector of GP-level characteristics,  $\alpha_{jk}$  is a country-level intercept term,  $w_j$  is a vector of country-level characteristics including distance measures,  $v_k$  is a company-level intercept term and  $z_k$  is a vector if company characteristics. The portfolio companies within countries represent the base-level observations, the countries and the GPs the upper-levels observations. Table 7 shows the explained variance in private equity returns, measured as the deal level PME, across the studied GP groups.

#### [Table 7 about here]

A multilevel linear model specification comes with a number of advantages in our setting. First, we are able to capture the GP-level (within country) relation between  $x_{ijkt}$  and performance in  $\beta$  by removing the country means of performance measures from all GP-level observations in  $x_{ijkt}$ . We include these means alongside the country-level variables to capture the country-level relation between  $w_i$  and the country intercept term  $\alpha_i$ in  $\gamma$ . We proceed the same way for portfolio companies by averaging their relevant variables by industry at the country level. Emerging markets are not a homogenous group and an EM region is often regarded as an already set diversified portfolio by investors (e.g. same continent countries such as Russia and China are not regarded the same although both Asian). By decomposing the GP-level variables in  $x_{ijt}$  into country means of returns and deviations from these means across GPs, then adding these means to the country characteristics in  $w_i$ , we are able to separate the within-country and across-country effects at the GP level (Bell and Jones, 2015). Furthermore, multilevel linear models correct for false positives in coefficient significance induced by pooled OLS. Indeed, varying sample sizes of GP investments across countries may falsely weigh in coefficients at the country level simply because of higher observation numbers, whereas multilevel linear models weigh in the coefficients by the precision of the sample observations at each level, which is for example inversely related to the its invested portfolio size within a country for the GP-level. Multilevel linear models are also better in correctly estimating and allow for better interpretation of interaction terms when using mean-centered independent variables (Aiken et al., 1991).

## 4. Descriptive statistics

Our data spans investments in both developed and emerging countries since 1930. In order to focus on the most significant period, we restrict the sample to investments starting from 1980 (years in which private equity institutionalized in developed markets) to 2010; and exited up to 2016<sup>13</sup>. Figure 2 shows fundraising and investment trends of private equity in developed and emerging markets with regards to the investment flow. Although allocation of private equity capital to emerging markets accounts for only 10% of global allocated capital in our sample period<sup>14</sup>, the invested capital in these markets represents 59% of capital inflows to these markets. That is, local GPs contribute to less than half the private capital on these markets, where foreign generalist funds count a 63% market share, alongside VC and Buyout funds which total 42% and 55% of invested capital over the sample period respectively.

#### [Figure 2 about here]

We clearly identify each private equity deal to each company it involves, the relevant fund's and GP's names and characteristics and their LPs. While 68% of funds in developed markets are organized into independent private partnerships in our sample, emerging markets funds are structured this way in only 44% of the cases, and about equally this figure into structures financed by DFIs. Both occurrences are observed for foreign

<sup>&</sup>lt;sup>13</sup> In our data, the maximum time to exit an investment on both markets averages 6 years (5,8 in median terms). Our investment period is then 1980-2010 and our performance measurement period is 1986-2016.

<sup>&</sup>lt;sup>14</sup> Accounting for all investment styles. On average, for illustrative purposes, 7% is allocated to VC funds and 6% to buyout funds in emerging markets across our sample period.

private equity investing in these markets with structures being 57% independent private partnerships and 23% being backed by DFIs.

We collect valuation data and performance metrics at the deal level, and timed cashflows at the fund level. Tables 2 to 6 highlight some of the characteristics of private equity investing and performance following the investment direction, figure 2 describes the evolution of raised and deployed private capital by origin and destination market, and figure 3 shows the distribution of return measures by deal and investment directions.

#### [Figure 3 about here]

Private equity in emerging markets is showing similar growth curves to that in developed markets, in terms of both fundraising and deployed capital. DMtoEM investors have higher-pace trajectories compared to pure EM-players, where growth trajectories are just shy of those observed in developed markets in the early 1980's. As shown in table 2, local emerging markets private equity only boomed in the late 1990's – early 2000's, although with some setbacks (EMtoEM), boosted by contribution from DM-based GPs.

Table 3 shows the average and median returns of private equity investments by style and GP groups. Almost all average and median GPs invested in both markets outperform pure local GPs on almost all styles and across different return measures. As highlighted in table 4, The same group of GPs also takes longer to invest in a portfolio company and has equivalent or higher holding periods than the average or typical local GP on both markets. Table 5 presents the frequency of exit routes for GP groups by investment and deal directions, where we observe higher proportions of trade sales in local developed markets (62%) compared to local emerging markets (41%). GPs investing in both markets fall in between with 57% exits in trade sales in developed markets versus 46% in emerging markets. IPOs are a dynamic exit strategy for local EM-based GPs with about half the total exits across the sample period, compared to only 26% in developed markets. Again, GPs with presence on both markets show higher proportions of IPO exits than their pure DM-based counterparts, with 31% and 41% proportions of exits in IPO on DMs and EMs respectively.

Fund and portfolio sizes are also significantly higher for the treatment group compared to locals. As shown in table 6, average buyout and venture capital funds are more than twice higher than those of pure DM or EM players, and portfolio sizes are also 40% to 80% higher for buyouts, almost equivalent on DM markets for VC investments but twice the size of portfolios in EMs. DMtoDM&EM GPs invest in younger buyout companies compared to the local GPs on average and to similarly aged companies for VC investments.

[Tables 2 to 6 about here]

# 5. Findings

In this section, we discuss the results of our estimation analysis with regards to the possible effects of geographical and cultural proximity on private equity returns

### 5.1. The geography of private equity performance

We estimate our multilevel linear model by adding the characteristics and level means progressively at each specification level and show the results in table 9.

#### [Table 9 about here]

Accounting for all GP, country and portfolio company characteristics, our results show persistent significance of the affiliation variable, highlighting a negative and significant association between being part of a local professional private equity organization and deal performance. This impact amounts to up to a - 1.5 impact on deal level PMEs compared to non-membership to a PE association, suggesting that deals that are concluded via this channel (networks amongst those associations notably) are not quality deals. Another possible explanation is that less experienced or lower skilled GPs which fail to spot interesting deals by their own means and networks, source lower quality deals within PE associations. This effect is negative and significant for DMs and EMs alike, but is insignificant for GPs investing on both markets within their EM portfolio. Likewise, Investment and exit speeds (time to exit and time to invest) are only relevant to pure

players and to the DM investments of the treatment group. Longer exit times are associated with up to a -0.48 impact on deal-level PME for each additional holding year in developed markets, both for pure players and GPs with investments on both markets. in emerging markets, longer holding periods do not affect performance either for local or foreign GPs. Similar conclusions can be drawn for investment speed (time to invest) with the same significance level but in opposite signs, as higher investment speed in developed markets is associated with higher returns in developed markets for both groups of DM-based GPs. There is no significant impact of higher investment speed on returns for DM-based GPs investing in emerging markets, unlike the local EM-based GPs for whom a one year waiting time to invest results in a significant 3% lower deal-level PME compared to the public benchmark.

Consistent with Gompers and Lerner (2000), at the country level, the money chasing hypothesis seems to hold in developed markets. The significant and negative effect of higher fund flows in matured developed markets results in higher valuations as a consequence of growing competition and scarcity of deals. This shows up in lower returns once the markets have cooled down by the time investments are exited. However, this effect is only observed in DM-based GPs investing in developed markets and is reversed for their EM holdings: a log dollar increase in fund inflows to emerging markets results in a positive and significant 0.7 higher PME compared to the public benchmark of where the GP is located, almost twice the similar observed effect observed for local emerging markets GPs.

Surprisingly, farther investment locations reflect positively in realized returns in DM-based GPs who are diversified geographically in developed markets. This result is inconsistent with what is documented in the asset management literature for example on hedge funds: geographically proximate hedge funds overperform geographically distant hedge funds (Teo, 2009).

Out of the four cultural dimensions, distance in hierarchy scores between the GP and target company countries reflect significantly in returns with the expected negative sign. Managers which are free from the order and complexity of hierarchical organizations in developed countries might be faced with the often rigid and inflexible nature of regulations within other developed countries. Alongside the hierarchy in administrative procedure and the general doing business environment, the effect is surprisingly reversed for local EM GPs, suggesting that sister EM-countries with higher managerial or administrative order and complexity welcome the free values of less hierarchical management.

At the portfolio company level, higher deal sequences are associated with significantly higher returns in emerging markets, with the effect being twice as important for locals compared to GPs investing in both DMs and EMs despite its small magnitude (0.003 and 0.006 increase in Deal-level PME from a deal to the next for the treatment group in emerging markets compared to local GPs respectively). For pure player DM-based GPs, higher deal sequences are associated with lower returns. Investing in companies which never had prior PE-backing for the first time negatively impacts GPs in DMs, the opposite expected effect shows in the club deal variable, suggesting that the risk of investing in a private portfolio company is often mitigated by risk-sharing between the GPs in a club deal. Unlike what has been documented in Choe et al. (2005), we generally do not see evidence of local emerging markets investors overperforming foreign investors.

5.1.1. Are there any crossed geographical and cultural effects between the GPs and the investment countries?

We introduce two interaction terms at the country level of GP characteristics with possible accentuated effect on the observed differences across groups of GPs based on their geographical investment focus. The first term is the Hirschman-Herfindahl measure for the GP's geographical concentration in a given country prior to the time of the investment relative to other countries, in terms of invested capital. This measure is considered high (respectively low) when the GP's capital allocation to companies in a considered country exceeds (respectively falls below) 50% of the relevant fund size. The second term is the cultural proximity of individuals within the investment teams of the GP. We use three measures to assess the effect of growing individual cultural proximity at the time of the deal: (i) a (static) dummy variable for whether the GP's investment teams count a speaker of the target country's language, (ii) a level variable for the number of language proficient teams within the investment team of the GP at the time of the deal, and (iii) a concentration measure of language proficient individuals in the investment teams of the GPs.

We interact those variables respectively with the geographical measure at the country level and investigate how higher previous geographical concentration of the GP interacts with a follow-on investment in the same geography, and how do culturally proximate individuals within the GP's investment team may close possible gaps or difficulties in deal making, occasioned by geographically distant investments. Results are given in table 10.

[Table 10 about here]

With regards to pure geographical interaction terms, investing in nearby countries with previous high GP presence is only relevant to foreign GPs on emerging markets and local EM-based GPs, where positive and significant effects on performance are noted. The opposite effect (low concentration in follow-on nearby invested countries) is reversely negative and significant, highlighting a possible shortfall from missing investment opportunities in nearby emerging markets destinations. In distant investment locations, overweighing capital allocation while being previously heavily invested in those locations results in significantly negative returns for pure DM-based players and foreign GPs investing in emerging markets, while the opposite effect is only noted for DM-invested GPs.

Simultaneous cultural and geographical distance interactions do not seem to affect returns within developed markets countries, suggesting more harmony and homogeneity in cultural values in these countries. For GPs present on both markets and local emerging markets GPs, having culturally closer investment teams in distant investment destinations positively and significantly reflects in realized returns.

### 5.1.2. Do private equity returns change geographically over time?

Because private equity investments are highly cyclical (Robinson and Sensoy, 2016), and as markets mature, we re-run our regressions by differentiating investments by sub-periods. Following table 2, and in order to focus on significant investment periods for both EM and DM markets, we split the sample period to investments prior to 2000 and those after 2000 (included).

#### [Table 13 about here]

The results show consistency in the observed effect of affiliation to a local private equity association. Interestingly, having prior EM investment experience positively and significantly reflects on returns after 2000 for GPs invested on both markets, but only for subsequent DM deals. Similar conclusions to the baseline estimation results can be drawn for the 2000-forward period, where most of the previously highlighted effects are noted, suggesting that those effects came with the maturing private equity markets.

#### 5.1.3. Does change in investment style when targeting emerging markets affect performance?

Given the previously highlighted cultural differences between countries, chances are GPs may deviate from their historically observed investment style and deal structures to better facilitate deal flow on culturally distant investment destinations. To this end, we use a style shift measure that is equal to one if the deal structure is the least observed in what is known of the GP in their investment history. Table 14 shows the results of the multilevel model estimates on this measure controlling for GP-, country- and portfolio company characteristics.

### [Table 14 about here]

Shifting from known deal structures to GPs reflects negatively and significantly in deal-level PMEs and multiples of invested capital, but offers the GP better chances of exiting via IPO or M&A. This might be

explained by the lack of GP experience in structuring new deal forms, which might result in either a pressure to successfully exit the investment or higher demand for this new deal structure among follow-on buyers.

### 5.2. The geography of private equity performance and return predictability

In this section, we examine whether success on a previous deal in a geography conditions success in the follow-on deal in the same geography. As this is often studied in the private equity literature under the term *persistence* in private equity returns, we are careful in using this terminology as our data are combined from several datasets which, even given their good quality as discussed previously, may not contain the full sequence of private equity deals for a given GP in a geography. Therefore, we analyze the performance of observed deal sequences in our dataset (sorted by investment dates, controlling for relevant fund vintages), and use the switch from a market to another in the DMtoDM&EM GP subgroup as robustness. Figure 4 illustrates the deal sequences in our sample by investment direction and deal direction.

#### [Figure 4 about here]

Table 11 shows the multilevel linear model estimates of the deal-level PME returns on the lagged deal-level PME returns by investment and deal directions. The results point out a significant positive relation between the previous deal return and the follow-on deal return for all GP groups in all markets. Assuming a causal relation, for DM-based GPs investing in DMs, a 1% higher return on the previous deal (relative to the public benchmark) is associated with 0.5 to 0.6 increase in the following deal in sequence, controlling for GP-, country- and company-level characteristics. The effect is significantly higher for EM-based GPs (0.7 higher return on the next deal for a 1% increase in the previous deal return) and interesting for DM-based GPs

investing in EMs, who have similar predictions to pure players in DMs even with lower predictions in the next returns compared to locals in EMs.

#### [Table 11 about here]

With regards to the effects of other GP-, country-, and portfolio company-level characteristics, they globally remain the same compared to the main specification estimates apart from a few changes. At the GP level and for the local EM-GPs, successful previous deals relax the investment speed (time to invest) on the follow-on deals and alleviate its significance along with the club deals indicator. This might be related to a lower need for a GP to compete or co-invest with another GP on a deal once they have a track-record for success on previous deals. At the country level, success on previous deals also minimizes the impact of the *money chasing deals* effect in emerging markets and further suppresses the significance in hierarchy distance. A possible explanation might be that previous deals build the experience and reputation of a given EM GP in ways that allows them to invest in other emerging markets without having to abide by the hierarchy of those markets each time or incur the effect of overvalued investments from flooding fund flows into those markets. At the portfolio company level, success in previous realized deals alleviates the significance of the negative impact associated with investing in venture capital deals in developed markets for GPs investing in both markets, but also significantly and positively weigh in the impact of higher deal sequence on overall follow-on returns for DMtoDM&EM GPs.

As pointed out earlier, because we may not have the full sequence of deals for each GP, we control our results using the DMtoDM&EM GP subsample. One reason why GPs list on databases is to generate deal

flow, which is often associated with tendency to list only successful deals or funds on commercial databases for investors to see. We argue however that there is a little chance for GPs to list deals with attention paid to how subsequent (or previous) deals on different markets and different fund vintages compare. To this end, we rearrange the deal order by GP in the DMtoDM&EM group by sequence within a fund in a market and we keep the deal associated with the first observed investment date. A fund is considered EM or DMfocused based on the highest frequency observed for either EM or DM within its portfolio companies' locations. We focus on first investments as they may reflect for either EMs or DMs a pressure to spend capital (Arcot et al., 2014). Then, we sort the funds on their vintage years in the fund family of each GP. We assign a sequence number for first investments this way and re-run the regressions of Deal-level PMEs on lagged deal-level PME using these new sequence numbers (See figure 5 for a visual).

#### [Figure 5 about here]

Our results are consistent with the previously documented positive effect and therefore assert for a positive relation between previous and next deal performance.

[Table 12 about here]

### 6. Conclusion

This paper investigates the performance drivers of private equity investing in emerging markets. Using a uniquely structured dataset and novel data, we show that cultural and geographical effects especially shape the investment directions of private equity capital flows into emerging markets, alongside previously documented performance drivers in the private equity literature. This effect is especially true for GPs investing in both markets compared to pure DM- and EM-players respectively. Our results are more consistent in the post-2000 investment period, and show predictability in returns along deal sequences on both markets and using different return measures. Cross-cultural and geographical effects are enhanced when the at the individual level for GPs, where investment teams are also culturally close using language as an indicator for cultural values. As the drawn conclusions can not be further extended to the entire global population of GPs, given that multilinear model estimates can only be interpreted within the studied groups and levels, we work in future versions on using Bayesian hierarchical modelling and other frequentist multilevel modelling techniques to extend the findings to larger populations of private equity GPs.

# 7. References

Aiken, L. S., and Stephen G. W.t, 1991. Multiple Regression: Testing and Interpreting Interactions, Sage: Newbury Park, CA.

Ain Tommar, S. and Darolles, S., (2017) Permanent Capital, Permanent Struggle? New Evidence from Listed Private Equity (Working paper). <u>https://papers.ssrn.com/sol3/papers.cfm?abstract\_id=2981021</u>

Arcot, S., Fluck, Z., Gaspar, J-M., and Hege, U., Fund Managers Under Pressure: Rationale and Determinants of Secondary Buyouts (2014). Journal of Financial Economics.

Bae, K-H, Stulz, R. and Tan, H., (2008) Do local analysts know more? A cross-country study of the performance of local analysts and foreign analysts, Journal of Fianncial Economics, vol. 88, issue 3, 581-606

Bell, A. and Jones, K., (2015), Explaining Fixed Effects: Random Effects Modeling of Time-Series Cross Sectional and Panel Data, Political Science Research and Methods, 3, issue 01, p. 133-153

Braun, R., Jenkinson, T., and Schemmerl, C, (2016), Adverse Selection and the Performance of Private Equity Co-Investments (working paper)

Braun, R., Jenckinson T. and Stoff, I., (2017), How Persistent is Private Equity Performance? Evidence From Deal Level Data, Journal of Financial Economics, vol. 123, issue 2, pages 273-291.

Charvel, R. (2012). The Roadmap for Private Equity, Venture Capital, and Hedge Funds in Mexico: Alternative Assets as an Accelerator of Credit to Firms in Emerging Markets. *Journal Of Private Equity*, 15(2), 53-62.

Chen, H., Gompers, P., Kovner, A., and Lerner, J., (2010), Buy local? The geography of venture capital, Journal of Urban Economics, Elsevier, vol. 67(1), 90-102

Choe H., Kho B-C., and Stulz, R. M., (2005). Do Domestic Investors Have an Edge? The Trading Experience of Foreign Investors in Korea, Review of Financial Studies, Oxford University Press for Society for Financial Studies, vol. 18(3), pages 795-829

Coval, J.D., and Moskowitz, T.J., (2001), The Geography of Investment: Informed Trading and Asset Prices, Journal of Political Economy, vol. 109, issue 4, 811-841

Coval, J.D., and Moskowitz, T.J., (1999), Home Bias at Home: Local Equity Preference in Domestic Portfolios, Journal of Finance, vol. 54, issue 6, 2045-2073

Gohil, R. (2014). The Performance of Private Equity Funds in India. The Journal Of Private Equity, (1), 63.

Gompers, P. and Lerner, J. (1998). Money Chasing Deals? The Impact of Fund Inflows on Private Equity Valuation. Journal of Financial Economics. 55. 281-325. 10.1016/S0304-405X(99)00052-5.

Harris, R. S., Jenkinson, T. and Kaplan, S. N. (2015), How Do Private Equity Investments Perform Compared to Public Equity?, Journal of Investment Management, forthcoming.

Harris, R. S., Jenkinson, T. and Kaplan, S. N. (2014), Private Equity Performance: What Do We Know? The Journal of Finance, 69: 1851–1882. doi:10.1111/jofi.12154

Hau, H., (2005), Location Matter: An examination of trading profits, Journal of Finance, Vol. 56, issue 5, 1959-1983.

Hochberg, Y., Ljungqvist, A., and Y. Lu, 2007, Venture Capital Networks and Investment Performance, Journal of Finance, 62: 251–301. doi:10.1111/j.1540-6261.2007.01207.x

Hofstede, G.H., 2001. Culture's Consequences: Comparing Values, Behaviors, Institutions, and Organizations across Nations (2nd ed.), Thousand Oaks, CA: SAGE Publications, Inc.

House, R. J., Hanges, P. J., Javidan, M., Dorfman, P. W., and Gupta, V. (Eds.). (2004). Culture, leadership, and organizations: The GLOBE study of 62 societies. Sage Publications.

Johan, S., & Zhang, M. (2016). Private equity exits in emerging markets. *Emerging Markets Review*, 29, 133-153. doi:10.1016/j.ememar.2016.08.016

Kaplan, S. N., and Strömberg, P., (2009). Leveraged Buyouts and Private Equity. Journal of Economic Perspectives, 23(1): 121-46.

Kaplan, S. N., and Schoar A., (2005). Private Equity Performance, Returns, Persistence, and Capital Flows. Journal of Finance Vol. 60, 1791–1823

Karolyi, G. A. (2016); Home Bias, an Academic Puzzle, Review of Finance, Volume 20, Issue 6, Pages 2049–2078, <u>https://doi.org/10.1093/rof/rfw007</u>

Korteweg, A. and Nagel, S. (2016), Risk-Adjusting the Returns to Venture Capital. The Journal of Finance, 71: 1437–1470. doi:10.1111/jofi.12390

La Porta, Rafael, Florencio Lopez de Silanes, Andrei Shleifer, and Robert W. Vishny, 1997, Trust in large organizations, American Economic Review Paper and Proceedings 87, 333–338.

Lerner, J., and Baker, M., (2017) An Empirical Analysis of Investment Return Dispersion in Emerging Market Private Equity, The Journal of Private Equity

L'Her J-F., Stoyanova R., Shaw, K., Scott, W., and Lai, C., (2016), A Bottom-Up Approach to the Risk-Adjusted Performance of the Buyout Fund Market, Financial Analysts Journal, Vol. 72, No. 4, 1–13

La Porta, Rafael, Florencio Lopez-de-Silanes, Andrei Shleifer and Robert W. Vishny, 2012. Legal determinants of external finance, Journal of Finance 52, 1131-1150.

Licht, A.N., C. Goldschmidt, and S.H. Schwartz, 2005. Culture, law, and corporate governance, International Review of Law and Economics 25, 229-255.

Licht, A.N., C. Goldschmidt, and S.H. Schwartz, 2007. Culture rules: The foundations of the rule of law and other norms of governance, Journal of Comparative Economics 35, 659-688.

Malloy, C.J., (2005), The geography of Equity Analysis, Journal of Finance, Vol. 60, issue 2, 719-755.

Minardi, A., Kanitz, R., & Bassani, R. (2014). Private Equity and Venture Capital In-dustry Performance in Brazil: 1990-2013. *The Journal Of Private Equity*, (4), 48.

Phalippou, L., and Gottschalg, O. (2009), The Performance of Private Equity Funds. The Review of Financial Studies, 22 (4): 1747-1776. doi: 10.1093/rfs/hhn014

Robinson, D. T., Sensoy, B. A. (2016), Cyclicality, Performance Measurement, and Cash-Flow Liquidity in Private Equity, Journal of Financial Economics, Volume 122, Issue 3, December 2016, Pages 521–543

Schwartz, S. H., (1994) Beyond individualism-collectivism: New dimensions of values., in U. Kim, H.C. Triandis, C. Kagitcibasi, S.C. Choi, and G. Yoon: Individualism and Collectivism: Theory, Method and Application, Sage, Newbury Park, CA.

Schwartz, S. H., (2004). Mapping and interpreting cultural differences around the world. In H. Vinkins, J. Soeters, and P. Esters: Comparing cultures: Dimensions of culture in a comparative perspective. Boston, MA: Brill Academic Publishers.

Shleifer, A., and Vishny, R. W., (1997), A survey of corporate governance, Journal of Finance, 52, 737-783

Teo, M. (2009), The Geography of Hedge Funds, The Review of Financial Studies, Volume 22, Issue 9, Pages 3531–3561, https://doi.org/10.1093/rfs/hhp007

Study	Sample size	Time period	Performance measures	Main findings
Panel A: At the fund le	vel			
Robinson and Sensoy (2016)	Data on 837 funds from one large LP	1984-2008	PME and tailored PME <sup>15</sup>	Private equity performance is cyclical. Funds raised in boom times underperform funds raised in bad times. Fund investors earn a liquidity premium in bad times.
<u>Harris, Jenckinson and</u> <u>Kaplan (2015)</u>	781 US buyouts invested by 300 LPs + 300 European buyouts	1984-2010	PME	Private equity funds outperform the S&P500 and is persistent in time. However, private equity performance is declining: net outperformance before 2006, but performance became roughly equal to that of the S&P500 from 2006 onward.
<u>Harris, Jenckinson and</u> <u>Kaplan (2014)</u>	1,400 US buyouts and VC funds invested by 200 LPs	1984-2008	PME	
<u>Phalippou and</u> <u>Gottschalg (2009)</u>	1,345 funds	1980-1993	Profitability Index (PI), Adjusted IRR, and Portion of investments that are successfully exited through an IPO or a sale to another company	Private equity's superior performance documented in previous studies drops to -3.83% per annum compared to the S&P500 after correcting for data bias.
<u>Kaplan and Schoar</u> (2005)	1,841 funds	1980-1997	IRR and PME	Returns net-of-fees to private equity investors are equal to the S&P500. Performance is persistent in time and is cyclical, with top performing funds being less sensitive to cyclicality effects.
Panel B: At the firm lev	vel			
Braun, Jenckinson and Stoff (2017)	Data derived from three large fund-of-fund managers: 13,523 portfolio company investments by 865 buyout funds	1974-2010	GPME <sup>16</sup>	Performance of private equity persistence has significantly declined as the industry has matured and competition grew for interesting deals.
<u>L'Her, Stoyanova,</u> <u>Shaw, Scott, and Lai</u> <u>(2016)</u>	Company data invested by 906 US buyout funds	1986-2014	Tailored PME	Private equity performance is consistent with previous literature findings using the PME, but private equity funds fail to outperform the market using tailored PME.
<u>Kaplan and Stromberg</u> (2009)	17,171 worldwide leveraged buyout transactions	1985-2007	Vintage year return, and annual capital commitment to U.S. private equity funds as a fraction of the U.S. stock market	Private equity fund returns tend to decline with increasing capital commitments, and capital commitments decline when realized returns decline
<u>Hochberg, Ljunqvist</u> and Lu (2007)	3,469 VC funds managed by 1,974 VC firms, involving 16,315 portfolio companies	1980-1999	Portion of investments that are successfully exited through an IPO or a sale to another company	Better-networked VC firms have better performance, and portfolio companies of better- networked VCs are significantly more likely to survive after the exit.

<sup>&</sup>lt;sup>15</sup> Kaplan and Schoar's (2005) Public Market Equivalent. It compares the return on the invested capital for private equity to what the investors would have earned for the same invested amount in the S&P500. Tailored PME is calculated using other public benchmarks. Tailored PME compares private equity performance to that of other market indices of publicly-traded companies which are similar to those invested by private equity funds.

<sup>&</sup>lt;sup>16</sup> Generalized PME, Korteweg and Nagel (2016)

### Appendix 1: Variable definition

Performance Metrics           Deal-level PME         Tailored public mader equivalent calculated at the deal level using regional MSCI indices of where the GP is board (North America, humpe, Asi, Hacfie and Emerging Markus).           Multiple of Invested Capital         Toral processifies from existind invested capital, scaled by find size.           Fist Rate         Number of existed investments to out any users to exist any scale and the existed investments.           GP-level Characteristics         Cop of each deal's relevant find size.           Indicator variable for whether the GP is affitiated with a local private equity organization (professional association or network of professionals).         Indicator variable for whether the GP is affitiated with a local private equity organization (professional association or network of professionals).           GP Direct Investment         Indicator variable for whether the GP is affitiated with a local private equity organization (professional association).           GP Direct Investment         Indicator variable for whether the GP iffered the deal for co-investment date.           Time to Invest Comparative Characteristies         The difference between the company's serie date and the prefexant find's first capital call date.           Capital Influence         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country. Comparative border.           Capital Influence         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund sec anative border.	Variable	Definition
Deal-level PME         Tailored public market capavalent calculated at the deal keel using regional MSC indices of where the GP is located (North America, Laroyce, Asia, Padic and Emerging Markets).           Multiple of Invested Capial         Total proceeds from caited investments to total invested capital, scaled by final size.           Fixit Rate         Number of exied investments to total invested capital, scaled by final size.           GP-level Characteristics         Size           Local Affiliation         Indicator variable for whether the GP is affiliated with a local private equity organization (professional association or a ctwork of professional association or a ctwork of the CP is the country of the country of the country of the difference between the Company's intervent data.           Copin Control Control Control Control Control Control Contrevanible for whether the CP's HQ country and the inves	Performance Metrics	
Indicate Capital         Invested Capital           Number of exited investments to train investment sto train investments in contain investment in contains investments in contains investment in contains investment in contains investment in the IP.           OP Direct Investment         Indicator variable for whether the GP infraced the deal for co-investment with the IP.           OP invest Investment         Indicator variable for whether the GP infraced the deal for co-investment with the IP.           OP invest Investment         Indicator variable for whether the GP infraced the deal for co-investment with the IP.           Optication Invest         The difference between the company's exit date and its investment date.           Time to Invest         The difference between the company's exit date and its investment date.           Control Characteristics         Capital Inflow           Capital Inflow         Log of total exited to private equity funds during the vintage year of each deal's relevant fund by country and the invested company's HQ country and the anne.           Nearby         Indicator variable for whether the GP's HQ country and the invested company's HQ country and an another boorder.           Calmad Distance sing Subwork? Calmad maked, sill sum and adaly for 80 anatris from Suban H SubmarY is probit           Calma	Deal-level PME	Tailored public market equivalent calculated at the deal level using regional MSCI indices of where the GP is
Multiple of Invested Capital         Total proceeds from exited investments to total invested capital, scale by fund size. <b>CP-level Characteristics</b> Version of the observation of the observatio		located (North America, Europe, Asia, Pacific and Emerging Markets).
Exit Rate         Number of exited investments by way of IPO or Trade Sale to total exited investments.           OP-level Characteristics         Log of each deal's relevant fund size.           Size         Log of each deal's relevant fund size.           Ord 101         Indicator variable for whether the CP is athilated with a local private capity organization (professional association or network of professional).           OP Direct Investment         Indicator variable for whether the CP of fired the deal for co-investment with the LP.           Der.EM Experience         Number of dals in the CP read-eccord prior to its firet EM investment date.           Time to Divest         The difference between the company's exit date and its investment date.           Conjunctive Vet Characteristics         Capital Inflow         Log of total equital allocated to private equity funds during the vintage year of each deal's relevant fund by country are the same.           Nearby         Indicator variable for whether the GP's HQ country and the invested company's HQ country and the invested company's HQ country and the invested company's HQ country and an or a marine border.           Calmed Distance         Statisting the social odeal, or available for whether the GP's HQ country and the invested company's HQ country and the invested company's HQ country also the orbito firm shallow H 5 dawarfy whether           Characteristics         Statisting the social odeal, or available for whether the GP's HQ country and the invested company's HQ country and the invested company's resperecting HQ country and the invested company's HQ country	Multiple of Invested Capital	Total proceeds from exited investments to total invested capital, scaled by fund size.
OP-level Characteristics         Log of each deal's relevant fund size.           Local Affiliation         Indicator variable for whether the GP is affiliated with a local private equity organization (professional association or network of professional).           GP Direct Investment         Indicator variable for whether the GP if effect the deal for on-investment with the LP.           New FN Experience         Number of deals in the GP track-record prior to its fort EM investment date.           Time to Exit         The difference between the company's investment date and is investment date.           Control-tevel Characteristics         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.           Cognital Diffuse         Indicator variable for whether the GP's IIQ country and the invested company's IIQ country and and or a maintime border.           Logal         Indicator variable for whether the GP's IIQ country and the invested company's IIQ country and and or a maintime border.           Columnt Zisteary         Columnt Zisteary and allocated or private equity funds during change and reatining tradition.           Embeddedness         Statust for Statust Zisteary in the Statust II Statust Zisteary in the Statust II Statust Zisteary in the statust and the invested company's IIQ country and and or a maintime border.           Columnt Zisteary         Coustratifie is the antary distance, measured is the diff	Exit Rate	Number of exited investments by way of IPO or Trade Sale to total exited investments.
Size         Log of each deal's relevant fund size.           Local Affiliation         Indicator variable for whether the GP is affiliated with a local private equity organization (professional association or network of professional).           GP Direct Investment         Indicator variable for whether the GP is affiliated with a local private equity organization (professional association or network of professional).           GP Direct Investment         Indicator variable for whether the GP offered the deal for co-investment date.           Time to Direct         The difference between the company's investment date.           Constructoristics         Constructoristics           Capital Inflow         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.           Congraphia         Indicator variable for whether the GP's HQ country and the invested company's HQ country and and in a marine border.           Distant         Indicator variable for whether the GP's HQ country and the invested company's HQ country and a marine border.           Calmed Diatane         Indicator variable for whether the GP's HQ country: and the invested company's HQ country and a marine border.           Calmed Diatane aning Schwartz's Calmed mold, and brain sevas: anallable for 8D examine from Shalon H Schwartz's whether           Embeddedness         Sustaining the social order, of avoiding change and retaining traditor.           Mastery         Sustaining the social order, of avoiding change and retaining traditor. </td <td>GP-level Characteristics</td> <td></td>	GP-level Characteristics	
Local Affiliation       Indicator variable for whether the QP is affiliated with a local private equity organization (professional association or network of professionals).         GP Direct Investment       Indicator variable for whether the QP effect the deal for on investment with the LP.         Pre-EM Experience       Number of deals in the QP track record prior to its first EM investment date.         Time to Invest       The difference between the company's investment date and its investment date.         Time to Invest       The difference between the company's investment date and the relevant fund's first capital call date.         Control_level Characteristies       The difference between the GP's 11Q country and the invested company's 11Q country are the same.         Local       Indicator variable for whether the GP's 11Q country and the invested company's 11Q country are a land or a maritime border.         Nearby       Indicator variable for whether the GP's 11Q country and the invested company's 11Q country are the same.         Indicator variable for whether the GP's 11Q country and the invested company's 1Q country are a land or a maritime border.         Indicator variable for whether the GP's 11Q country and the invested company's 1Q country are the same.         Indicator variable in the master of a sound gename and retaining tradition.         Our variable in the master of a sound gename and retaining tradition.         Our variable in the master of sound gename and retaining tradition.         Our variable is the master of slistone, massured as the differen	Size	Log of each deal's relevant fund size.
OP Direct Investment       Indicator variable for whether the GP offered the deal for 0-investment with the LP.         Co-investment       Indicator variable for whether the GP offered the deal for 0-investment with the LP.         Pre-EXI Experience       Number of deals in the GP indic-record prior to its first EXI investment date.         Time to Invest       The difference between the company's investment date.         Control-level Characteristics       Control-level Characteristics         Copyrability Distance       Indicator variable for whether the GP's HQ country and the invested company's HQ country are the same.         Indicator variable for whether the GP's HQ country and the invested company's HQ country are a land or a maritime border.       Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.         Calural Distance       Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.         Calural Distance       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness statuse, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the heinerky distance, measured as the difference between the insterly scores of the GP's and company's respective HQ countries.         Hierarchy       Existe	Local Affiliation	Indicator variable for whether the GP is affiliated with a local private equity organization (professional association
GP Direct Investment         Indicator variable for whether the GP offered the del for co-investment with the LP.           Or-investment         Indicator variable for whether the GP offered the del for co-investment with the LP.           Pre-EM Experience         Number of dels in the GP tack-record prior to its first EAI investment date.           Time to Dist         The difference between the company's circuit date and its investment date.           Contryl-teel Characteristics         Contryl-teel Characteristics           Contryl-teel Characteristics         Indicator variable for whether the GP's IIQ country and the invested company's IIQ country are the same.           Indicator variable for whether the GP's IIQ country and the invested company's HQ country share a land or a maritime border.         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Calland Distance using Schwards' Cultural model, with using anilable for 80 countries firm Scholm H Schwards' archite         Embeddedness           Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Success through individual personal action (as opposed to group action).           Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.           Mastery         Success through individual personal action (as opposed to group action). <td< td=""><td></td><td>or network of professionals).</td></td<>		or network of professionals).
Co-investment       Indicator variable for whether the GP offered the deal for co-investment with the LP.         Pne-EM Experience       Number of deals in the GP rack-record prior to is first EM investment date.         Time to Exit       The difference between the company's exit date and its investment date.         Capital Inflow       Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.         Cogenphaled Distance       Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Local       Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Columnt Distance using Schwart's (Culumd medic, with sorce andicable for 80 nontrins from Shalem H Schwart's verbite         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Hierarchy       Success through individual personal action (as opposed to group action).         Our variable is the meastery distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of class roal order, with people in superior positions and others in inferior positions.         Our variable is the hierarchy distance, measured as the difference between the hierarchy	GP Direct Investment	Indicator variable for whether the deal is a direct investment by the GP.
Pre-EM Experience         Number of deals in the GP rack-record prior to is first EM investment date.           Time to Dist         The difference between the company's civat data and is investment date.           Capital Inflow         Edifference between the company's investment date and the relevant fund's first capital call date.           Capital Inflow         Indicator variable for whether the GP's HQ country and the invested company's HQ country are the same.           Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Cultural Distance using Schwartz's Catural model, with cores multible for 80 matrix fing Staburt H Schwartz's newlite.         Embeddedness           Cultural Distance using Schwartz's Catural model, with cores multible for 80 matrix fing Staburt H Schwartz's newlite.         Embeddedness           Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Suscess through individual personal action (ac opposed to group action).           Our variable is the interperity distance, measured as the difference between the interfere positions.           Our variable is the interperity distance, measured as the difference between the interfere positions.           Our variable is the interept distance, m	Co-investment	Indicator variable for whether the GP offered the deal for co-investment with the LP.
Time to Isix       The difference between the company's exi date and its investment date.         Time to Invest       The difference between the company's investment date and the relevant fund's first capital all date.         Capital Inflow       Log of total capital allocated to private equipy funds during the vintage year of each deal's relevant fund by country.         Compublical Distance       Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Distant       Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Cultural Distance using Schwart's Cultural model, mits ourse usailable for 8D countries from Shulum 11 Schwart's orbite!         Embeddedeness       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedeness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Our variable is the mastery distance, measured as the difference between the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Indicator variable for whether the company is appective IQ countries.         Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Intelectual Autonomy       Indicator va	Pre-EM Experience	Number of deals in the GP track-record prior to its first EM investment date.
Time to lavest         The difference between the company's investment date and the relevant fund's first capital call date.           Constry-level Characteristics         Log of total capital allocated to private capity funds during the vintage year of each deal's relevant fund by country.           Gengraphical Diatana         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Distant         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Calland Distance using Schwart?'s Calland model, with sore: analithe for 80 matrics from Shahm H Schwart?' website           Embeddedness         Sustaining the social order, of avoiding change and retaining tradition.           Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Sustance of the GP's and company's respective HQ countries.           Mastery         Existence of clear social order, with people in superior positions and others in inferior positions. Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.           Hierarchy         Existence of the GP's and company's respective HQ countries.           Corpany-level Characteristics         Indicator variable for whether the company has never been previously PE-backed           First Time Deal         Indicator varia	Time to Exit	The difference between the company's exit date and its investment date.
Country-level Characteristics         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.           Grappiphial Distance         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Local         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Cultural Distance using Schwart?'s Cultural works which constructs from Shahan H1 Schwart?'s website         Cultural Distance using Schwart?'s cultural works which constructs from Shahan H1 Schwart?'s website           Embeddedness         Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Suscess through individual personal action (as opposed to group action).         Our variable is the hierarchy distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Intellectual Autonomy         Indicator variable for whether the de only is respective HQ countries.           Intellectual Autonomy         Indicator variable for whether the de and thoughts (individual, political, etc.)           Our variable is the intellectual autonomy scores of the GP's ind company's respective HQ countries.           First Time Deal         Indicator variable for whether the deal is invested by more than one PE frm	Time to Invest	The difference between the company's investment date and the relevant fund's first capital call date.
Capital Inflow         Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.           Gegraphical Distance         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Nearby         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Caluard Distance using Schwart?; Cultural model, with source analable for 8D countries from Shalom H Schwart?; whole         Embeddedness           Embeddedness         Sustaining the social order, of avoiding change and retaining radition.           Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Success through individual personal action (as opposed to group action).           Our variable is the mastery distance, measured as the difference between the interfers positions.           Our variable is the interactry distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.           Hiterarchy         Existence of clear social order, with people in superior positions and others in inferior positions.           Our variable is the interlectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.           Indicator variable for whether the company has never been previously PE-backed	Country-level Characteristics	
Coordination         Indicator variable for whether the GP's HQ country and the invested company's HQ country set the same. Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border. Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.           Calural Distance using Schwart's Cultural model, with some available for 80 annitris from Shalum H Schwart's vehict         Cultural Distance using Schwart's Cultural model, with some available for 80 annitris from Shalum H Schwart's vehict           Embeddedness         Sustaining the social order, of avoiding change and retaining tradition. Our variable is the modeddeness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Success through individual personal action (as opposed to group action). Our variable is the meatery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.           Intellectual Autonomy         Indecator variable for whether the company has never been previously PE-backed           Indicator variable for whether the deal is venture Capital         And ofference between the intellectual autonomy scores of the GP's and company's respective HQ countries.           Indicator variable for whether the company has never been previously PE-backed         Indicator variable for whether the company has never been previously PE-backed           Indicator variable for whether the deal is invested by more than one PE firm         Indidicator variable for whether the deal is invested by more than o	Capital Inflow	Log of total capital allocated to private equity funds during the vintage year of each deal's relevant fund by country.
Local         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Distant         Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.           Caltural Distance using Schwartg's Caltural mold, with scores available for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.           Caltural Distance using Schwartg's Caltural mold, with scores available for 80 countries from Shalom H Schwartg's website           Embeddedness         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.           Mastery         Success through individual personal action (as opposed to group action). Our variable is the mastery distance, measured as the difference between the interfor positions. Our variable is the interactly dual to distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.           Hierarchy         Existence of clar social order, of autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.           Cotomany         Indicator variable for whether the company has never been previously PE-backed           Indicator variable for whether the deal is invested by more than one PE firm           Our variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least obsereted deal structure characte	Geographical Distance	
Nearby       Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a maritime border.         Distant       Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.         Caltural Distance using Schwartg's Caltural model, wills cores and silds for 80 countries from Shalom H Schwartg's website         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action).         Our variable is the imastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions.         Our variable is the interarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Thelectual Autonomy       Indicator variable for whether the company has never been previously PE-backed         VC Durny       Indicator variable for whether the deal is invested by more than one PE frm         Our variable for whether the deal is invested by more than one PE frm         Other variables       Indicator variable for whether the deal is invested by more than one PE frm	Local	Indicator variable for whether the GP's HQ country and the invested company's HQ country are the same.
marine border.       Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.         Cultural Distance using Schwartz's Cultural model, with scores available for 80 contries from Shahm H Schwartz's website         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition. Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action). Our variable is the heirastry distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions. Our variable is the insteady distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.) Our variable is the instellectual autonomy distance, measured as the difference between the intellectual autonomy across of the GP's and company's respective HQ countries.         Commany       Indicator variable for whether the ceal is Venture Capital         Age       Log of invested capital in deal         VC Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the deal is invested dorist compared the GP's historical deals. <td>Nearby</td> <td>Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a</td>	Nearby	Indicator variable for whether the GP's HQ country and the invested company's HQ country share a land or a
Distant       Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land or a maritime border.         Calutual Distance using Schwartz's Calutual model, with some available for 80 countries from Shalom H Schwartz's mebrite         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action).         Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions.         Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.)         Our variable for whether the deal is Venture Capital       Age         Age       Log of romyster clapital         Oug of invested capital in deal       Nontexter the deal is Venture Capital         Age       Log of company age at financing in years         Invested Capital       Log of orwaster capital in deal         Deal Sequence       The Sequence of the deal in t		maritime border.
Cathreal Distance using Schwartz's Cathreal model, with scores available for 80 countries from Shalom H Schwartz's website         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition. Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action). Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Company-level Characteristics       Indicator variable for whether the company has never been previously PE-backed         First Time Deal       Log of invested capital deal         VC Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, et	Distant	Indicator variable for whether the GP's HQ country and the invested company's HQ country do not share a land
Cultural Distance using Schwart?'s Cultural model, with scores available for 80 countries from Shalom H Schwart?'s website         Embeddedness       Sustaining the social order, of avoiding change and retaining tradition. Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective IIQ countries.         Mastery       Success through individual personal action (as opposed to group action). Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective IQ countries.         Hierarchy       Existence of clar social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective IQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective IQ countries.         Company-level Characteristics       Indicator variable for whether the deal is Venture Capital Age         VC Dummy       Indicator variable for whether the deal is Venture Capital Age         Log of invested capital in deal       Log of invested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake, vs. majority stake, etc.) are the least observed deal structure char		or a maritime border.
Embeddedness       Sustaining the social order, of avoiding change and retaining tradition.         Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action).         Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions.         Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and company's respective HQ countries.         First Time Deal       Indicator variable for whether the company has never been previously PE-backed         YC Dummy       Indicator variable for whether the deal is Venture Capital         Age       Log of invested capital in deal         Invested Capital       Log of invested capital in deal         Other variables       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for Whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed	Cultural Distance using Schwartz's Cultural	! model, with scores available for 80 countries from Shalom H Schwartz's website
Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the GP's and company's respective IIQ countries.MasterySuccess through individual personal action (as opposed to group action). Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective IIQ countries.HierarchyExistence of clear social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective IIQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective IIQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backed Log of company age at financing in years Invested Capital Log of invested capital in deal Deal SequenceDeal SequenceThe Sequence of the deal in the GP fund program Indicator variable for whether the deal is invested by more than one PE firmOther variableIndicator variable for whether the company's investment style characteristics (e.g., VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g., VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g., VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g., VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure character	Embeddedness	Sustaining the social order, of avoiding change and retaining tradition.
GP's and company's respective HQ countries.         Mastery       Success through individual personal action (as opposed to group action).         Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions.         Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.)         Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Company-level Characteristics       First Time Deal         VC Dummy       Indicator variable for whether the company has never been previously PE-backed         VC Dummy       Indicator variable for whether the deal is Venture Capital         Age       Log of rompany age at financing in years         Invested Capital       Log of novested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, et		Our variable is the embeddedness distance, measured as the difference between the Embeddedness scores of the
Mastery       Success through individual personal action (as opposed to group action).         Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.         Hierarchy       Existence of clear social order, with people in superior positions and others in inferior positions.         Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.         Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.)         Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Company-level Characteristics       Indicator variable for whether the company has never been previously PE-backed         First Time Deal       Indicator variable for whether the deal is Venture Capital         Age       Log of company age at financing in years         Invested Capital       Log of invested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a count		GP's and company's respective HQ countries.
Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and company's respective HQ countries.HierarchyExistence of clear social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backedFirst Time DealIndicator variable for whether the deal is Venture Capital AgeInvested CapitalLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics (respectively Link) when the GP capital allocation to companies in the considered outry relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation firvate equity capital. The measure is considered High (respectively Low) when the GP capital allocated private equity capital. The measur	Mastery	Success through individual personal action (as opposed to group action).
Hierarchycompany's respective HQ countries.HierarchyExistence of clear social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backed Indicator variable for whether the deal is Venture Capital Log of company age at financing in years Invested CapitalNew VC DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) wheth the GP has an investment professional who speaks the language of the target company's country at the time of the deal.HI GP Geo ConcentrationIndicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.I anguage Speakers DummyIndicator variable for whether th		Our variable is the mastery distance, measured as the difference between the mastery scores of the GP's and
HierarchyExistence of clear social order, with people in superior positions and others in inferior positions. Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backed Indicator variable for whether the deal is Venture Capital Age Log of company age at financing in years Indicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the deal is investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level and proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment swere made.GP-level andural proximity measuresIndicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.HH GP Geo ConcentrationHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in <b< td=""><td></td><td>company's respective HQ countries.</td></b<>		company's respective HQ countries.
Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and company's respective HQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backed Indicator variable for whether the deal is Venture Capital Log of company age at financing in years Log of company age at financing in years Indicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level cultural proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.GP-level cultural proximity measuresIndicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.HIH GP Geo ConcentrationIndicator variable for whether the GP has an investment professional who speaks the language of the tar	Hierarchy	Existence of clear social order, with people in superior positions and others in inferior positions.
Intellectual Autonomycompany's respective HQ countries.Intellectual AutonomyIndependence and openness of ideas and thoughts (individual, political, etc.) Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backedFirst Time DealIndicator variable for whether the deal is Venture CapitalAgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.Inaguage SpeakersNumber of investment professional in the investment professional who speaks the language of the target company's country at the time of the deal.HIH GP Geo ConcentrationIndicator variable for whether the GP has an investment professional who speaks the language of the target 		Our variable is the hierarchy distance, measured as the difference between the hierarchy scores of the GP's and
Intellectual Autonomy       Independence and openness of ideas and thoughts (individual, political, etc.)         Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.         Company-level Characteristics       Indicator variable for whether the company has never been previously PE-backed         First Time Deal       Indicator variable for whether the company has never been previously PE-backed         VC Dummy       Indicator variable for whether the deal is Venture Capital         Age       Log of company age at financing in years         Invested Capital       Log of invested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       HIrischman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment swere made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of th		company's respective HQ countries.
Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy scores of the GP's and company's respective HQ countries.Company-level CharacteristicsIndicator variable for whether the company has never been previously PE-backedFirst Time DealIndicator variable for whether the deal is Venture CapitalAgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.HH GP Geo ConcentrationHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.GP-level cultural proximity measuresIndicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.HIL Language SpeakersNumber of investment professionals in the investment teams of the	Intellectual Autonomy	Independence and openness of ideas and thoughts (individual, political, etc.)
scores of the GP's and company's respective HQ countries.         Company-level Characteristics         First Time Deal       Indicator variable for whether the company has never been previously PE-backed         VC Dummy       Indicator variable for whether the deal is Venture Capital         Age       Log of company age at financing in years         Invested Capital       Log of invested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers Dummy       Indicato		Our variable is the intellectual autonomy distance, measured as the difference between the intellectual autonomy
Company-level CharacteristicsFirst Time DealIndicator variable for whether the company has never been previously PE-backedVC DummyIndicator variable for whether the deal is Venture CapitalAgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.Language SpeakersNumber of investment professionals in the investment teams of the GP at the time of the deal.Language SpeakersNumber of investment professionals in the investment team of one form of the CP at the		scores of the GP's and company's respective HQ countries.
First Time DealIndicator variable for whether the company has never been previously PE-backedVC DummyIndicator variable for whether the deal is Venture CapitalAgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.GP-level cultural proximity measuresIndicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.Language SpeakersNumber of investment professionals in the investment teams of the GP at the time of the deal.HUL Language SpeakersHurde provement team of the CD at the	Company-level Characteristics	
VC DummyIndicator variable for whether the deal is Venture CapitalAgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.Language SpeakersNumber of investment professionals in the investment teams of the GP at the time of the deal.	First Time Deal	Indicator variable for whether the company has never been previously PE-backed
AgeLog of company age at financing in yearsInvested CapitalLog of invested capital in dealDeal SequenceThe Sequence of the deal in the GP fund programClub Deal DummyIndicator variable for whether the deal is invested by more than one PE firmOther variablesIndicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.GP-level geographical proximity measuresHirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investment professional who speaks the language of the target company's country at the time of the deal.Language SpeakersNumber of investment professionals in the investment teams of the GP at the time of the deal.HIM Language SpeakersNumber of investment professionals in the investment teams of the GP at the time of the deal.	VC Dummy	Indicator variable for whether the deal is Venture Capital
Invested Capital       Log of invested capital in deal         Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals. <i>GP-level geographical proximity measures</i> Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively Low) stake below) 50% of the fund size through which investments were made. <i>GP-level cultural proximity measures</i> Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HIH L organgers       Number of investment professionals in the investment team of the deal.	Age	Log of company age at financing in years
Deal Sequence       The Sequence of the deal in the GP fund program         Club Deal Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         Hurd Language Speakers       Hirefordahl measure for country at the time or the deal.	Invested Capital	Log of invested capital in deal
Club Deal Dummy       Indicator variable for whether the deal is invested by more than one PE firm         Other variables       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.	Deal Sequence	The Sequence of the deal in the GP fund program
Other variables         Style Shift       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         Hull Language Speakers       Hirschman Herfindahl measure for country and the investment teams of the GP at the investment of the deal.	Club Deal Dummy	Indicator variable for whether the deal is invested by more than one PE firm
Style Shift       Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs. majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         GP-level geographical proximity measures       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HIL Language Speakers       Hirschman Herfindahl measure for culturely ervine on the deal.	Other variables	
GP-level geographical proximity measures       majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.         HH GP Geo Concentration       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HH Language Speakers       Hirschman Herfindahl measure for country at the time of the deal.	Style Shift	Indicator variable for whether the company's investment style characteristics (e.g. VC vs. Buyout, minority stake vs.
GP-terel geographical proximity measures         HH GP Geo Concentration       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.		majority stake, etc.) are the least observed deal structure characteristics compared the GP's historical deals.
HH GP Geo Concentration       Hirschman-Herfindahl measure for a GP's geographical concentration in a country relative to other countries in the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High (respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively falls below) 50% of the fund size through which investments were made.         GP-level cultural proximity measures       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HH Language Speakers       Hirschman Herfindahl measure for culturally environ to a company and the investment teams of the GP at the investment of the deal.	GP-level geographical proximity measures	
Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.	HH GP Geo Concentration	Hirschman-Hertindahl measure for a GP's geographical concentration in a country relative to other countries in
GP-level cultural proximity measures         Ianguage Speakers         Language Speakers         UHL Language Speakers		the same group (EM or DM), in terms of total allocated private equity capital. The measure is considered High
GP-level cultural proximity measures         Language Speakers         Language Speakers         UHL Language Speakers		(respectively Low) when the GP capital allocation to companies in the considered country exceeds (respectively
CP-tevel cultural proximity measures         Language Speakers Dummy         Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HH Language Speakers       Hirschman Harfindel measure for culturally account on the investment team of the CP at	CD hard antennal target in	raus below) 50% of the fund size through which investments were made.
Language Speakers Dummy       Indicator variable for whether the GP has an investment professional who speaks the language of the target company's country at the time of the deal.         Language Speakers       Number of investment professionals in the investment teams of the GP at the time of the deal.         HH Language Speakers       Hirschman Harfindel measure for culturally proving to professionals within the investment team of the CP at the	Gr-uvel cultural proximity measures	
Language Speakers CD at the company's country at the time of the deal. HIL Language Speakers Hirrschman Harfradel measure for culturally according to professionals within the investment team of the CD at the	Language Speakers Dummy	indicator variable for whether the GP has an investment professional who speaks the language of the target
Language Speakers INUmber of investment professionals in the investment teams of the GP at the time of the deal.	I	company's country at the time of the deal.
HEREBORDER HOPPERDOND CONTRACTOR AND THE PROPERDON AND THE PROPERO	Language Speakers	Inumber of investment professionals in the investment teams of the GP at the time of the deal.
time of the deal	IIII Language opeakers	time of the deal

**Figure 1: Sample Structure.** The figure shows the structure of our sample groups following investment and deal directions, and nesting properties by geography and culture. GPs and companies are either DM-based or EM-based following their respective headquarters locations. Following this classification, we first establish a *deal direction*, from the GP to the Portfolio company: DMtoDM, DMtoEM, EMtoDM and EMtoEM. We further aggregate the deal directions by GP to establish an *investment direction*, that is groups of GPs who are observed to either invest in DMs only, in EMs only, or in both (solid line groups in the figure). We have instances of DM-based GPs investing solely in EMs and of EM-based GPs investing solely in DMs (dashed line groups in the figure), but we do not consider those as they are not important in number or in invested capital, nor economically significant. Our distance measure is twofold: a geographical distance and a cultural distance. See appendix 1 for distance variable definitions.



**Figure 2: Private Equity Fundraising and Investing by Investment Direction.** The plot shows our aggregated database figures for raised and deployed capital by investment direction: *DMtoDM* denotes investments by DM-based GPs in developed markets, *DMtoEM* investments by DM-based GPs in emerging markets, *EMtoEM* investment by EM-based GPs in emerging markets, and *EMtoDM* investments by EM-based GPs in developed markets.





**Table 2 : Summary Statistics – Number of deals by Investment year, Investment Direction and Deal Direction.** The Deal Direction describes the GP investment flow based on where the GP and the target company are headquartered respectively (i.e. from a DM-based (resp. EM-based) GP to a DM-based (resp. EM-based) target company. The Investment Direction is established by aggregating the observed deal directions of each GP (i.e. GPs who invest in DMs only, in EM only, or in both).

	I	nvestment Direction	n and Deal Direction	n
Investment year	DMtoDM	DMtoD	M&EM	EMtoEM
		DMtoDM	DMtoEM	
Pre-1980	1 768	1 713	8	2
1980	586	562	1	1
1981	1 021	1 113	4	
1982	1 511	1 606	15	
1983	2 449	2 427	19	3
1984	2 880	2 673	12	
1985	2 664	2 451	33	
1986	2 966	2 879	27	
1987	3 050	2 987	32	2
1988	2 747	2 549	100	2
1989	2 644	2 761	67	8
1990	2 097	2 266	55	12
1991	1 797	1 846	10	23
1992	2 146	2 495	22	43
1993	1 799	2 225	36	50
1994	1 918	2 476	65	84
1995	2 213	2 902	116	107
1996	3 246	4 272	258	172
1997	4 197	4 939	229	174
1998	4 958	6 172	234	248
1999	7 189	9 277	348	597
2000	11 990	13 912	760	1 779
2001	8 616	8 679	459	820
2002	6 1 4 6	5 576	255	577
2003	6 850	6 225	340	712
2004	7 841	7 343	390	730
2005	7 404	7 069	505	797
2006	7 559	6 704	718	905
2007	8 315	7 103	887	1 243
2008	7 868	6 437	812	1 303
2009	3 1 3 0	2 569	223	486
2010	5 223	4 041	512	1 001
2011	3 649	3 260	291	529
2012	2 892	2 599	209	288
2013	2 049	1 885	124	153
2014	1 243	1 227	106	126
2015	839	841	76	86
2016	353	315	34	35
Total	147 813	148 376	8 392	13 098

 Table 3 : Summary Statistics – Average and median returns by Investment Style, Investment Direction and Deal Direction.
 Only fully exited investments are taken into account. See appendix 1 for variable definitions.

Panel A : Deal-level PMEs, by Investment Style, Investment Direction and Deal Direction										
	Inves	stment Direction	n and Deal Dire	ection						
DMtoDM DMtoDM&EM EMtoEM Total										
		DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM			
Mean	4.63	6.50	4.47	3.67	5.30	4.47	3.67			
Median	4.00	7.22	4.27	3.77	5.68	4.27	3.77			
Mean	5.22	5.28	4.73	4.97	5.24	4.73	4.97			
Median	5.36	4.96	4.77	4.90	5.26	4.77	4.90			
Mean	4.99	3.92	5.35	4.06	4.51	5.35	4.06			
Median	4.36	2.81	5.75	5.21	2.88	5.75	5.21			
Mean	5.29	6.07	5.55	4.77	5.60	5.55	4.77			
Median	4.94	6.85	5.19	4.92	5.69	5.19	4.92			
Mean	6.00	6.57	3.50	5.25	6.23	3.50	5.25			
Median	5.95	7.14	1.15	5.83	7.00	1.15	5.83			
Mean	6.53	4.04	3.90	2.39	5.28	3.90	2.39			
Median	6.08	2.67	2.67	0.71	5.61	2.67	0.71			
Mean	5.10	5.64	4.79	4.78	5.29	4.79	4.78			
Median	5.21	5.75	4.77	4.84	5.40	4.77	4.84			
	Mean Median Median Median Median Median Median Median Median Median Mean Median Median	As, by Investment Style, InvestInvestDMtoDMMean4.63Median4.00Mean5.22Median5.36Mean4.99Median4.36Mean5.29Median4.94Mean6.00Median5.95Mean6.53Median6.08Mean5.10Median5.21	As, by Investment Style, Investment Direction           Investment Direction           DMtoDM         DMtoDM           Mean         4.63         6.50           Median         4.00         7.22           Mean         5.22         5.28           Median         5.36         4.96           Mean         4.36         2.81           Median         5.29         6.07           Median         5.95         7.14           Mean         6.03         4.04           Median         6.08         2.67           Median         6.08         2.67           Mean         6.10         5.64           Median         5.21         5.75	As, by Investment Style, Investment Direction and Deal Direction and Deal Direction         Investment Direction and Deal Direction           DMtoDM         DMtoDM&EM           DMtoDM         0MtoEM           Mean         4.63         6.50         4.47           Median         4.00         7.22         4.27           Mean         5.22         5.28         4.73           Median         5.36         4.96         4.77           Mean         4.36         2.81         5.75           Median         5.29         6.07         5.55           Median         4.94         6.85         5.19           Mean         6.00         6.57         3.50           Median         5.95         7.14         1.15           Mean         6.08         2.67         2.67           Mean         6.08         2.67         2.67           Median         5.10         5.64         4.79           Median         5.21         5.75         4.77	As, by Investment Style, Investment Direction and Deal Direction           Investment Direction and Deal Direction           DMtoDM         DMtoDM&EM         EMtoEM           DMtoDM         0MtoDM         0MtoEM           Mean         4.63         6.50         4.47         3.67           Median         4.00         7.22         4.27         3.77           Mean         5.22         5.28         4.73         4.97           Median         5.36         4.96         4.77         4.90           Mean         4.99         3.92         5.35         4.06           Median         4.36         2.81         5.75         5.21           Mean         5.29         6.07         5.55         4.77           Median         4.94         6.85         5.19         4.92           Mean         6.00         6.57         3.50         5.25           Median         5.95         7.14         1.15         5.83           Mean         6.08         2.67         2.67         0.71           Median         6.08         2.67         2.67         0.71           Mean         6.08         2.67         2.67         0	As, by Investment Style, Investment Direction and Deal Direction           Investment Direction and Deal Direction           DMtoDM         DMtoDM&EM         EMtoEM           DMtoDM         OMtoDM         OMtoDM           Mean         4.63         6.50         4.47         3.67         5.30           Median         4.00         7.22         4.27         3.77         5.68           Mean         5.22         5.28         4.73         4.97         5.24           Median         5.36         4.96         4.77         4.90         5.26           Mean         4.36         2.81         5.75         5.21         2.88           Mean         4.36         2.81         5.75         5.21         2.88           Mean         5.29         6.07         5.55         4.77         5.60           Median         4.94         6.85         5.19         4.92         5.69           Mean         6.00         6.57         3.50         5.25         6.23           Median         5.95         7.14         1.15         5.83         7.00           Mean         6.08         2.6	As, by Investment Style, Investment Direction and Deal Direction           Investment Direction and Deal Direction           DMtoDM         DMtoDM&EM         EMtoEM         Total           DMtoDM         DMtoDM         DMtoEM         DMtoDM         DMtoEM           Mean         4.63         6.50         4.47         3.67         5.30         4.47           Median         4.00         7.22         4.27         3.77         5.68         4.27           Mean         5.22         5.28         4.73         4.97         5.24         4.73           Median         5.36         4.96         4.77         4.90         5.26         4.77           Mean         4.36         2.81         5.75         5.21         2.88         5.75           Median         4.36         2.81         5.75         5.21         2.88         5.75           Median         4.94         6.85         5.19         4.92         5.69         5.19           Median         4.94         6.85         5.19         4.92         5.69         5.19           Mean         6.00         6.57         3.50         5.25         6.23         3.50 <td< td=""></td<>			

# Panel B : Value Multiples by Investment Style, Investment Direction and Deal Direction

		Inves	stment Direction	h and Deal Dire	ection	_		
		DMtoDM	DMtoD	M&EM	EMtoEM		Total	
Style			DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM
Burrout	Mean	2.14	2.70	2.65	1.59	2.43	2.65	1.59
Buyout	Median	1.62	2.10	2.33	0.95	Total           DMtoDM         DMtoEM           2.43         2.65           1.88         2.33           3.79         3.24           3.48         2.30           2.45         2.78           2.42         2.45           2.85         2.56           2.09         1.57           2.67         2.31           1.67         0.93           1.81         1.06           1.17         0.23           3.51         2.97           3.11         2.30	0.95	
Weinsteine Consistel	Mean	3.51	4.06	3.24	3.07	3.79	3.24	3.07
venture Capitai	Median	2.93	3.58	4.00 $5.24$ $5.07$ $5.79$ $3.58$ $2.30$ $2.82$ $3.48$ $2.66$ $2.78$ $3.61$ $2.45$ $3.64$ $2.45$ $3.02$ $2.42$ $2.97$ $2.56$ $2.12$ $2.85$	2.30	2.82		
Erred of Erredo	Mean	2.15	2.66	2.78	3.61	2.45	2.78	3.61
Fund of Funds	Median	1.38	3.64	2.45	3.02	2.42	2.45	3.02
Concerting Driverty Equity	Mean	2.59	2.97	2.56	2.12	2.85	2.56	2.12
Generalist Private Equity	Median	1.64	2.69	1.57	1.67	2.09	1.57	1.67
Manager	Mean	2.44	3.33	2.31	1.82	2.67	2.31	1.82
Mezzanine	Median	1.64	2.62	0.93	1.72	1.67	0.93	1.72
	Mean	2.10	1.49	1.06	1.96	1.81	1.06	1.96
Other Private Equity	Median	1.43	1.17	0.23	1.86	1.17	0.23	1.86
T-+-1	Mean	3.25	3.75	2.97	2.86	3.51	2.97	2.86
1 otal	Median	2.62	3.41	2.30	2.60	3.11	1.57 2.31 0.93 1.06 0.23 2.97 2.30	2.60

#### Panel C : Exit Rates by Investment Style, Investment Direction and Deal Direction

						<b>—</b>				
		DMtoDM	DMtoD	M&EM	EMtoEM	Total				
Style			DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM		
Purrout	Mean	0.30	0.36	0.30	0.13	0.33	0.30	0.13		
Buyout	Median	0.29	0.36	0.31	0.11	0.33	0.31	0.11		
Venture Copital	Mean	0.51	0.56	0.39	0.20	0.53	0.39	0.20		
Venture Capitai	Median	0.57	0.60	0.39	0.17	0.58	0.39	0.17		
Fund of Funds	Mean	0.30	0.37	0.26	0.23	0.34	0.26	0.23		
	Median	0.32	0.46	0.22	0.29	0.41	0.22	0.29		
Concerlict Driverty Estricts	Mean	0.30	0.42	0.32	0.21	0.38	0.32	0.21		
Generalist Private Equity	Median	0.27	0.43	0.29	0.19	0.39	0.29	0.19		
Magganina	Mean	0.34	0.30	0.14	0.19	0.33	0.14	0.19		
Mezzainne	Median	0.33	0.30	0.06	0.15	0.31	0.06	0.15		
Othon Brizzato Equity	Mean	0.33	0.34	0.31	0.12	0.33	0.31	0.12		
Other Private Equity	Median	0.32	0.35	0.31	0.02	0.34	0.31	0.02		
Tetel	Mean	0.46	0.51	0.35	0.20	0.49	0.35	0.20		
I otal	Median	0.51	0.54	0.33	0.17	0.53	0.33	0.17		

Table 4 : Summary Statistics – Average and median Investment and Holding Periods by Investment Style, Investment Direction and Deal Direction. Investment Period is the time in years between the fund first capital call date and the company investment date. Holding Period is the time in years that a GP takes to exit the investment, measured as the difference between the exit date and the company investment date.

		Invest	ment Direction	n and Deal Dire	ection				
		DMtoDM DMtoDM&EM		DM&EM	EMtoEM	Total			
Style			DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM	
Derrort	Mean	7.36	9.24	8.03	6.29	8.27	8.03	6.29	
Buyout	Median	4.00	4.00	3.04	2.92	4.00	3.04	2.92	
Venture Capital	Mean	6.51	8.03	8.72	5.85	7.26	8.72	5.85	
	Median	4.25	4.50	4.17	3.05	4.39	4.17	3.05	
Fund of Funds	Mean	4.83	4.99	4.46	3.18	4.93	4.46	3.18	
Fund of Funds	Median	3.00	3.33	2.41	2.06	3.22	2.41	2.06	
Conomitat Driverto Equity	Mean	11.84	18.11	19.91	11.43	15.91	19.91	11.43	
Generalist Private Equity	Median	8.29	15.76	13.50	4.57	13.33	13.50	4.57	
Managina	Mean	4.82	5.31	5.11	3.87	4.95	5.11	3.87	
Mezzahine	Median	3.50	3.94	3.09	2.58	3.60	3.09	2.58	
Other Directo Erritor	Mean	5.06	6.16	6.37	2.50	5.53	6.37	2.50	
Other Private Equity	Median	3.30	3.64	4.28	1.54	3.46	4.28	1.54	
T-4-1	Mean	6.87	9.17	10.58	6.37	8.02	10.58	6.37	
Total	Median	4.31	4.77	4.45	3.15	4.50	4.45	3.15	

Panel B : Holding Periods by Investment Style, Investment Direction and Deal Direction - in years

		Invest	ment Direction	n and Deal Dire	ection			
		DMtoDM	DMtoI	DM&EM	EMtoEM		Total	
Style			DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM
Purrout	Mean	5.78	6.11	5.40	4.11	5.94	5.40	4.11
Buyout	Median	5.10	5.40	5.10	3.80	5.20	5.10	3.80
Wantana Carital	Mean	6.64	6.56	5.89	4.44	6.60	5.89	4.44
Venture Capitai	Median	6.00	6.00	5.35	3.90	6.00	5.35	3.90
Englad Engla	Mean	6.96	6.70	6.17	4.59	6.81	6.17	4.59
Fund of Funds	Median	6.60	6.40	5.70	3.95	6.50	5.70	3.95
Concertiet Driverte Erreiter	Mean	6.18	6.27	4.86	4.92	6.24	4.86	4.92
Generalist Private Equity	Median	5.60	5.60	4.30	4.40	5.60	I otal           M         DMtoEM           5.40         5.10           5.89         5.35           6.17         5.70           4.86         4.30           4.67         4.50           4.43         3.95           5.53         5.10	4.40
Mananina	Mean	6.46	6.04	4.67	4.02	6.34	4.67	4.02
Mezzanine	Median	5.50	5.10	4.50	3.80	5.40	I otal           DMtoEM           5.40           5.10           5.89           5.35           6.17           5.70           4.86           4.30           4.67           4.50           4.43           3.95           5.53           5.10	3.80
	Mean	5.04	5.08	4.43	6.12	5.06	4.43	6.12
Other Private Equity	Median	4.20	4.10	3.95	6.40	4.10	3.95	6.40
T-4-1	Mean	6.50	6.48	5.53	4.45	6.48	5.53	4.45
Total	Median	5.90	5.80	5.10	3.90	5.90	5.10	3.90

Table 5 : Summary Statistics - Exit Strategies by Investment Direction and Deal Direction.

-	DMtoDM	DMtoD	M&EM	EMtoEM		Total	
Exit Type		DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM
Buyback	0.01	0.00	0.01	0.01	0.00	0.01	0.01
IPO	0.26	0.31	0.41	0.49	0.29	0.41	0.49
Reverse Takeover	0.01	0.01	0.00	0.02	0.01	0.00	0.02
Secondary Sale	0.06	0.05	0.10	0.06	0.06	0.10	0.06
Trade Sale	0.62	0.57	0.46	0.41	0.59	0.46	0.41
Write Off	0.05	0.05	0.01	0.01	0.05	0.01	0.01

Table 6 : Summary Statistics – Average fund size (in USD millions), portfolio size (in number) and age of invested companies (in years), by Investment Direction and Deal Direction. This table shows sample average fund size, average portfolio size and average portfolio company age at financing by investment style, investment direction and deal direction. Variable definitions are detailed in appendix 1.

		DMtoDM	DMtoD	M&EM	EMtoEM		Total	
Investment Style			DMtoDM	DMtoEM		DMtoDM	DMtoEM	EMtoEM
	Fund Size	419.25	1143.37	1043.03	301.07	787.14	1043.03	301.07
Buyout	Portfolio Size	61.44	111.04	69.97	49.71	85.48	69.97	49.71
	Age of company at financing	20.66	18.76	16.66	21.84	19.73	16.66	21.84
	Fund Size	99.32	294.20	443.67	72.01	197.23	443.67	72.01
Venture Capital	Portfolio Size	216.88	292.79	431.01	240.60	254.17	431.01	240.60
	Age of company at financing	6.13	5.59	9.15	8.09	5.86	9.15	8.09
	Fund Size	285.76	637.14	438.22	81.21	501.31	438.22	81.21
Fund of Funds	Portfolio Size	53.17	72.85	36.74	59.08	64.66	36.74	59.08
	Age of company at financing	9.07	7.09	6.35	8.99	7.95	6.35	8.99
	Fund Size	151.94	3616.49	2754.92	156.39	2294.90	94.90 2754.92	156.39
Generalist Private Equity	Portfolio Size	129.39	406.03	292.36	88.78	307.27	292.36	88.78
	Age of company at financing	13.34	11.23	13.26	14.94	11.98	13.26	14.94
	Fund Size	218.33	542.74	387.65	220.07	296.82	387.65	220.07
Mezzanine	Portfolio Size	167.07	373.71	205.76	1138.10	220.19	205.76	1138.10
	Age of company at financing	17.64	22.91	14.98	12.70	19.02	14.98	12.70
	Fund Size	456.27	1413.39	1063.58	194.84	884.85	1063.58	194.84
Other Private Equity	Portfolio Size	20.84	28.24	23.41	9.81	24.02	23.41	9.81
	Age of company at financing	11.72	13.01	10.70	13.61	12.28	10.70	13.61
	Fund Size	146.63	571.96	799.96	100.62	361.47	799.96	100.62
Total	Portfolio Size	187.24	278.31	326.00	214.27	232.87	326.00	214.27
	Age of company at financing	8.66	7.88	11.31	9.95	8.26	11.31	9.95



**Table 7 : Percentage of explained variance in returns by level.** This table shows the percentage of explained variance pertaining to each data level using ANOVA variance decomposition. Return measures are detailed in appendix 1.

	DMtoDM DMtoDM&EM H			EMtoEM					
					Overall				
	GP	Country	Company	GP	Country	Company	GP	Country	Company
Exit Rate	17%	19%	64%	9%	23%	67%	1%	72%	27%
Value Return	43%	9%	48%	30%	19%	51%	8%	29%	63%
Deal-level PME	9%	11%	80%	1%	13%	86%	6%	16%	78%
	DMtoDM								
Exit Rate				22%	15%	63%			
Value Return				53%	5%	42%			
Deal-level PME				7%	20%	73%			
					DMtoEM				
Exit Rate				22%	16%	62%			
Value Return				13%	58%	59%			
Deal-level PME				3%	22%	75%			

Table 8 : Correlation tables of GP, Country and Company level characteristics. This table shows the correlations between the GP-level characteristics, the Country-level characteristics, and the Company-level characteristics. Variable definitions are given in Appendix 1.

Panel A : Correlations b	Panel A : Correlations between GP-level characteristics										
	Size	Local Affiliation	GP Direct Investment	Co-investment	Pre-EM Experience	Time to Exit	Time to Invest				
Size	1.0000										
Local Affiliation	0.2874	1.0000									
GP Direct Investment	0.0472	0.0221	1.0000								
Co-investment	0.0488	0.0207	-0.0160	1.0000							
Pre-EM Experience	0.2176	0.3919	-0.0278	-0.0283	1.0000						
Time to Exit	-0.0181	-0.0110	-0.0054	-0.0032	0.0129	1.0000					
Time to Invest	-0.0203	0.0139	0.1618	0.1443	0.0433	0.0678	1.0000				
Panel B : Correlations b	etween Country-	level characteristic	s and GP mean	s of Country-level	l variables						
	Capital Inflow	Nearby	Distant	Embeddedness distance	Hierarchy distance	Mastery distance	Intellectual autonomy distance	GP Mean Exit Rate	GP Mean Multiple of Invested Capital	GP Mean Deal PME	
Capital Inflow	1.0000										
Nearby	-0.0408	1.0000									
Distant	-0.0972	-0.0631	1.0000								
Embeddedness distance	-0.0140	0.2743	0.1384	1.0000							
Hierarchy distance	0.0406	0.1677	0.1830	0.5059	1.0000						
Mastery distance	-0.0683	0.1416	-0.0159	0.5027	-0.3867	1.0000					
Intellectual autonomy distance	-0.0628	0.0264	-0.0710	-0.0618	-0.7273	0.7407	1.0000				
GP mean Exit Rate	0.1725	-0.1904	-0.2288	-0.1012	-0.1040	-0.0029	0.0729	1.0000			
GP mean Multiple of Invested Capital	-0.0106	-0.0990	-0.1187	-0.0746	-0.0889	0.0051	0.0509	0.1587	1.0000		
GP mean Deal PME	0.0206	-0.1917	-0.2594	-0.1725	-0.1830	-0.0141	0.0767	0.2756	0.9312	1.0000	
Panel C : Correlations b	etween Compan	y-level Characteris	tics and Countr	y means of Comp	any-level variable	es (by company i	industry)				
	First Time PE deal	VC dummy	LN Age at financing	LN invested capital	Deal Sequence	Club deal dummy	Country mean Exit Rate	Country mean Multiple of invested capital	Country mean Deal PME		

	deal	VC dummy	financing	capital	Deal Sequence	dummy	Exit Rate	Multiple of invested capital	Deal PME	
First Time PE deal	1.0000									
VC dummy	-0.0009	1.0000								
LN Age at financing	-0.0023	-0.2666	1.0000							
LN invested capital	-0.1037	-0.0138	-0.0163	1.0000						
Deal Sequence	-0.1307	0.0538	-0.0488	0.0895	1.0000					
Club deal dummy	-0.0125	0.1546	-0.0888	0.3532	0.0400	1.0000				
Country mean Exit Rate	0.0063	0.2209	-0.1619	0.2333	0.1021	0.2366	1.0000			
Country mean Multiple of invested capital	0.0090	0.2631	-0.1760	0.2088	0.1034	0.2084	0.7670	1.0000		
Country mean Deal PME	0.0127	0.2262	-0.1639	0.2359	0.1118	0.2293	0.9126	0.8751	1.0000	

Table 9 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

	Within GPs			Across Countries				Across Companies							
	DMtoDM	D	MtoDM&E	М	EMtoEM	DMtoDM	DMtoDM&EM		EMtoEM	DMtoDM	oDM DMtoDM&EM		М	EMtoEM	
GP Characteristics	-	Overall	DMtoDM	DMtoEM		-	Overall	DMtoDM	DMtoEM		-	Overall	DMtoDM	DMtoEM	
Size	0.0959	-0.0362	-0.0544	0.546***	-0.0275	0.178	0.131	0.116	0.297	-0.131*	0.244*	0.0648	0.0478	0.107	-0.117
	(1.06)	(-0.54)	(-0.79)	(3.34)	(-0.63)	(1.67)	(1.79)	(1.56)	(1.43)	(-2.16)	(2.11)	(0.83)	(0.60)	(0.49)	(-1.87)
Local Affiliation	-1.367***	-1.635***	-1.688***	0.140	-0.429**	-1.459***	-1.491***	-1.538***	0.348	-0.485**	-1.511***	-1.525***	-1.555***	-0.129	-0.901***
	(-4.45)	(-6.11)	(-6.18)	(0.19)	(-3.16)	(-4.25)	(-5.29)	(-5.37)	(0.39)	(-2.63)	(-4.16)	(-5.21)	(-5.23)	(-0.14)	(-4.50)
GP Direct Investment	-0.780	-0.388	-0.459	1.361	0.164	-0.815	-0.393	-0.446	1.412	0.356	-0.716	-0.584	-0.604	1.439	0.274
	(-0.87)	(-0.53)	(-0.61)	(1.00)	(0.59)	(-0.80)	(-0.50)	(-0.55)	(0.87)	(0.93)	(-0.68)	(-0.72)	(-0.73)	(0.92)	(0.69)
Co-investment	-0.474	-0.460	-0.488	0.472	0.419	-0.481	-0.380	-0.415	1.106	0.518	-0.328	-0.461	-0.486	1.322	0.534
	(-0.54)	(-0.61)	(-0.63)	(0.34)	(1.54)	(-0.48)	(-0.47)	(-0.50)	(0.68)	(1.40)	(-0.32)	(-0.56)	(-0.58)	(0.83)	(1.38)
Pre-EM Experience		0.000153	0.000162	-0.0516**			0.000211	0.000243	-0.0438*			-0.000153	-0.0000166	1.057	
		(1.07)	(1.12)	(-2.89)			(1.40)	(1.59)	(-2.21)			(-0.58)	(-0.06)	(0.50)	
Time to Exit	-0.394***	-0.274***	-0.277***	-0.148	-0.00955	-0.443***	-0.303***	-0.300***	-0.194	-0.0241	-0.483***	-0.305***	-0.298***	-0.165	-0.0317
	(-6.28)	(-4.64)	(-4.48)	(-1.33)	(-0.36)	(-5.93)	(-4.65)	(-4.43)	(-1.34)	(-0.66)	(-5.95)	(-4.32)	(-4.06)	(-1.09)	(-0.82)
Time to Invest	0.151***	0.143***	0.145***	0.0227	-0.0115	0.163***	0.153***	0.156***	0.00500	-0.0247*	0.165***	0.140***	0.142***	-0.00550	-0.0268*
	(7.66)	(9.24)	(9.23)	(0.56)	(-1.43)	(7.54)	(9.46)	(9.48)	(0.11)	(-2.32)	(7.16)	(8.04)	(8.05)	(-0.12)	(-2.43)
Country Characteristics								0.00.000							
Capital Inflow						-0.0276	-0.757***	-0.804***	0.507	0.111	0.226	-0.657***	-0.696***	0.717*	0.262*
						(-0.21)	(-6.8/)	(-7.13)	(1.82)	(1.61)	(1.48)	(-5.19)	(-5.39)	(2.18)	(2.54)
Nearby						1.582	3.698***	3.711***	1.023	-2.031	1.076	4.151***	4.421***	1.761	-1.968
D' + +						(1.17)	(3.79)	(3.72)	(0.28)	(-1.59)	(0.69)	(3.91)	(4.06)	(0.50)	(-1.51)
Distant						(1.20)	1.704**	1.781*	-1.443	0.0568	0.694	2.111**	Z.280**	-0.331	0.0200
Embaddada a distance						(1.20)	(2.59)	(2.55)	(-1.35)	(0.03)	(0.78)	(2.97)	(3.03)	(-0.31)	(0.02)
Embeddedness distance						(0.24)	(1.15)	(1.27)	-0.643	0.103	(0.29)	(1.432	(1.12)	-0.06/9	(0.300
Hiorarchy Distance						(0.24)	(1.13) <b>2 217**</b>	(1.27)	(-0.33)	(0.10) 1 100**	(0.38)	(1.40) 2 602***	(1.1 <i>2)</i> 3.015*	(-0.04)	(0.55) 1 205**
Therateny Distance						(1.24)	(310)	(2.52)	(1.15)	(2.73)	-1.780	(3.33)	(2.47)	(1.00)	(2.80)
Mastery Distance						(-1.24)	2 308	0.280	4 726	0.164	0.472	(-5.55)	(-2.+7)	3.614	(2.0)
Mastery Distance						(0.20)	(0.88)	(0.06)	(-0.70)	(-0.15)	(-0.07)	(0.41)	(0.24)	(-0.54)	(-0.51)
Int Autonomy Distance						-3.176	-2 097	-1 700	-1 115	0.460	-1.839	-1 989	-2 447	-2 296	0.713
Int. Hatohomy Distance						(-1 44)	(-1.83)	(-1.25)	(-0.42)	(0.41)	(-0.74)	(-1 54)	(-1.58)	(-0.87)	(0.64)
Company Characteristics						( 111)	(1100)	(1120)	( 0.12)	(0.11)	( 011 1)	(1101)	(1100)	( 0.07)	(0.01)
First Time Investment											-0.829**	-0.802**	-0.812**	-0.0743	0.269
											(-2.64)	(-2.68)	(-2.67)	(-0.08)	(1.46)
VC Dummy											0.484	-0.795*	-0.834*	-0.240	-0.0133
5											(0.97)	(-2.10)	(-2.15)	(-0.29)	(-0.05)
Age											0.0895***	0.149***	0.157***	-0.0194	0.0116
Ŭ,											(3.95)	(7.55)	(7.69)	(-0.43)	(1.03)
Invested Capital											-1.076***	-1.201***	-1.297***	-1.005*	-0.189
-											(-4.46)	(-5.34)	(-5.54)	(-2.36)	(-1.74)
Deal Sequence											-0.0150***	0.000706	0.000503	0.00346***	0.00611***
											(-7.99)	(1.67)	(1.14)	(4.35)	(5.22)
Club Deal Dummy											2.169	2.717*	3.102*	1.489	0.648*
											(1.69)	(2.35)	(2.53)	(0.92)	(2.01)
Intercept											Yes	Yes	Yes	Yes	Yes
Investment year FE											Yes	Yes	Yes	Yes	Yes
Industry FE											Yes	Yes	Yes	Yes	Yes
Number of Countries											26	47	25	22	20
Number of Companies											9 677	9 885	9 437	448	687
Number of observations											44 351	53 578	52 523	1 055	1 353

Table 10 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME, with cross-level effects of Distance measures at the country and GP levels. This table shows the interaction between the geographic distance of the company's country from the GP's country and the intensity of cultural proximity of the GP investment teams on the one hand, and the interaction between the geographic distance of the company country on the other hand. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

	DMtoDM	DMtoDM&EM			EMtoEM
		Overall	DMtoDM	DMtoEM	-
	170	NTC .	N/TPO	170	N/FIG
GP Characteristics	YES	YES	YES	YES	YES
Country Characteristics	YES	YES	YES	YES	YES
Cross-level interactions					
GP geographical concentration vs. investment country geographical distance					
Nearby x HH GP Geo Concentration Low	0.58	1.42	-0.26	-0.68*	-1.02*
	(1.22)	(0.89)	(-0.68)	(-1.99)	(-2.20)
Nearby x HH GP Geo Concentration High	0.49	0.94**	0.48	1.59**	1.03**
	(0.56)	(2.83)	(1.04)	(2.53)	(2.82)
Distant x HH GP Geo Concentration Low	0.78*	1.02**	0.52**	0.84	0.74
	(2.03)	(2.65)	(2.81)	(0.21)	(0.08)
Distant x HH GP Geo Concentration High	-0.54*	0.24	-0.65	-0.52*	-0.41
	(-2.33)	(1.23)	(-0.01)	(-2.37)	(-0.89)
GP cultural proximity vs. investment country geographical distance					
Nearby x Lang. Speakers Dummy	-1.367	1.02*	-0.071	1.56***	1.20**
	(-1.72)	(2.02)	(-0.01)	(4.86)	(2.79)
Distant x Lang. Speakers Dummy	-0.091	0.31*	0.305	1.183	0.43**
	(-1.37)	(2.36)	(0.33)	(1.13)	(2.52)
Nearby x Lang. Speakers	-0.017	2.03*	2.01***	1.06***	1.08**
	(-0.15)	(2.49)	(6.23)	(5.21)	(2.67)
Distant x Lang. Speakers	-0.027	1.02*	1.24***	0.98***	1.02**
	(-0.35)	(2.39)	(5.37)	(4.03)	(2.68)
Nearby x HH Lang. Speakers	-1.01	1.85***	1.74***	1.86***	2.65***
	(-1.84)	(3.44)	(3.90)	(7.40)	(4.16)
Distant x HH Lang. Speakers	-1.06	1.52***	2.05**	1.41***	1.35***
	(-0.49)	(3.06)	(2.39)	(6.66)	(3.06)
Company Characteristics	YES	YES	YES	YES	YES
Intercept	Yes	Yes	Yes	Yes	Yes
Investment year FE	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes
Number of Countries	26	47	25	22	20
Number of Companies	9 677	9 885	9 437	448	687
Number of observations	44 351	53 578	52 523	1 055	1 353

Figure 4: Deal Sequences sorted on GPs and investment year by investment and deal directions. This figure shows the sequence number of the sample deals by GP group and investment year.





3000

Deal Sequence 1000 2000

0

1970





1990 Investment year

1980

DMtoEM investments - DMtoDM&EM GPs

.

2010

2000



Table 11 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME, accounting for previous deal performance on follow-on deals performance. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

Overall         DMtoDM         DMtoEM           Lagged Deal-level PME         0.575***         0.551***         0.249***         0.755***           Size         0.00432         4.00540         4.00669         0.114         -00169           Size         0.00432         4.00540         4.00669         0.0141         -0.0169           Local Affiliation         -0.6869**         -0.822***         -0.2415         -0.341***           GP Direct Investment         -0.535         -0.563         -0.573         1.054         -0.00154           GP Direct Investment         -0.111         -0.239         -0.255         0.676         0.0184           GP Direct Investment         -0.0111         -0.239         -0.0255         0.676         0.0184           Pre-EM Experience         -0.000147         -0.000355         1.574         -0.00174         -0.029           Time to Invest         0.109***         0.00927***         0.0029         0.00149         -0.0129           Contry Characteristics         -0.279***         0.019***         0.0027***         0.0029         0.00149           Contry Characteristics         -0.117         -0.271**         -0.240**         0.0131         0.0331           Contry Characteris		DMtoDM		DMtoDM&EM		EMtoEM
Lagged Deal-level PME         0.575***         0.551***         0.551***         0.249***         0.755***           GP Characteristics         (13.03)         (12.02)         (13.13)         (12.02)         (13.27)         (47.64)           Size         0.00432         -4.0540         -4.0660         0.114         -0.0169           Local Affiliation         -0.689**         -0.822***         -0.245         -0.341***           GP Direct Investment         -0.353         -0.563         -0.573         1.054         -0.00147           Co-investment         -0.111         -0.239         -0.255         0.676         0.0184           Co-investment         -0.0161         -0.000045         1.574         -0.0219           Co-investment         -0.141***         -0.143**         -0.0103         -0.0219           Time to Exit         -0.279***         -0.141***         -0.143**         -0.0103         -0.0219           Time to Invest         0.110***         0.990***         0.9927***         0.0029         -0.00496           Camint Characteristics			Overall	DMtoDM	DMtoEM	
Characteristics         (19.09)         (13.63)         (12.02)         (13.27)         (47.64)           GP Characteristics         (0.05)         (-1.05)         (-1.28)         (0.54)         (-0.51)           Local Affiliation         -0.0669         -0.806***         -0.822***         (-0.24)         (-0.51)           Local Affiliation         (-0.75)         (-1.06)         (-1.06)         (-0.75)         (-0.06)         (-1.06)         (-0.27)           GP Direct Investment         (-0.15)         (-1.06)         (-1.06)         (-0.07)         (-0.00)           Co-investment         (-0.111)         -0.239         -0.255         0.676         0.0184           Time to Exit         (-0.279***         -0.000147         -0.0000545         1.574         (-0.021)           Time to Invest         0.110***         0.00906***         0.00927***         0.0029         -0.00496           Contry Characteristics         (-6.68)         (7.33)         (7.29)         (0.07)         (-0.82)           Contry Characteristics         (-1.11)         (-2.59)         (-2.81)         (-2.68)         (-0.0014)           Distant         -0.349         0.986*         1.140*         0.110         -0.028)           Distant<	Lagged Deal-level PME	0.575***	0.551***	0.551***	0.249***	0.755***
GP Characteristics         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C         C <thc< th="">         C         <thc< th=""></thc<></thc<>	00	(19.09)	(13.63)	(12.02)	(13.27)	(47.64)
Size $0.00432$ $-0.0540$ $-0.0669$ $0.114$ $-0.0169$ Local Affiliation $0.669^{0ers}$ $0.822^{erss}$ $0.0245^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.005^{orss}$ $0.025^{orss}$ $0.025^{orss}$ $0.0005^{orss}$ $0.025^{orss}$ $0.0025^{orss}$ $0.0025^{ors$	GP Characteristics		( /			
(0.05)         (-1.05)         (-1.28)         (0.54)         (-0.51)           Local Affiliation         -0.660***         -0.896***         -0.822***         -0.245         -0.241         *         -0.241         (-4.22)         (-6.22)         (-5.36)           GP Direct Investment         -0.153         -0.573         1.054         -0.00044         -0.00044           (-0.16)         (-1.06)         (-1.06)         (0.72)         (-0.02)         (-5.36)           Co-investment         -0.0111         -0.239         -0.255         0.076         0.0184           (-0.16)         (-0.44)         (-0.430)         (0.086)         -0.0219           Time to Exit         -0.279***         -0.0141***         -0.0143***         -0.0299         -0.04065           Contry Characteristics         (-1.01)         -0.217***         -0.0240***         0.00299         -0.04066           Contry Characteristics         (-1.04)         (-2.59)         (-2.81)         (-0.84)         (0.610)           Nearby         -1.068         1.629*         1.144*         0.0110         -0.034           Distant         -0.354         0.366*         1.140*         0.0110         -0.034           Distant         -0.052**	Size	0.00432	-0.0540	-0.0669	0.114	-0.0169
Local Affiliation         -0.680***         -0.806***         -0.822***         -0.241*5         -0.341***           GP Direct Investment         (-2.74)         (-4.20)         (-4.22)         (-0.27)         (-0.005)4           Co-investment         -0.111         -0.239         -0.255         0.676         0.0018           Co-investment         -0.111         -0.239         -0.255         0.676         0.009           Pre-EM Experience         -0.000147         -0.0000545         1.574         -0.0019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00019         -0.00029         -0.00049         -0.00029         -0.00049         -0.0029         -0.00049         -0.0029         -0.00049         -0.0029         -0.0049         -0.0029         -0.0049         -0.0029         -0.0049         -0.02019         -0.0029         -0.0049         -0.02019         -0.02019         -0.0029         -0.0049         -0.02019         -0.02019         -0.02019         -0.0029         -0.0049         -0.02019         -0.02019         -0.0029         -0.0049         -0.02019         -0.0131         <		(0.05)	(-1.05)	(-1.28)	(0.54)	(-0.51)
	Local Affiliation	-0.680**	-0.806***	-0.822***	-0.245	-0.341***
$ \begin{array}{c ccccccccccccccccccccccccccccccccccc$		(-2.74)	(-4.20)	(-4.22)	(-0.29)	(-3.36)
(40.75)         (-1.06)         (-0.72)         (-0.02)           Co-investment         -0.111         -0.239         -0.255         0.676         0.0184           (-0.16)         (-0.44)         (-0.45)         (0.09)         0.85         0.000147         -0.0000545         1.574           Time to Exit         -0.279***         -0.141***         -0.133         (-0.103         -0.0219           Time to Exit         -0.279***         0.0906***         0.0927***         0.0029         -0.0049           Time to Invest         0.110***         0.0906***         0.0927***         0.0049         -0.0049           Country Characteristics         (-1.04)         -0.217**         0.240**         0.84***         0.0017           Castal Inflow         -0.117         -0.217**         -0.240**         0.848**         0.0017           Castat         -0.349         0.986*         1.714*         2.271         -0.324           Castat         -0.034         0.986*         1.10         -0.034           Distaat         -0.058         (2.11)         (2.31)         (0.11)         -0.034           Berbeddedness distance         3.551         1.368         3.667         -1.0034         0.453         1.	GP Direct Investment	-0.535	-0.563	-0.573	1.054	-0.00504
$\begin{array}{ccc} \mbox{Co-investment} & -0.111 & -0.239 & -0.255 & 0.676 & 0.0184 \\ & (-0.16) & (-0.44) & (-0.46) & (-0.45) & (0.09) \\ \mbox{Pre-EM Experience} & -0.000147 & -0.0000545 & 1.574 & (-0.83) & (-0.30) & (-0.86) \\ \mbox{Time to Exit} & -0.279*** & -0.141*** & -0.143** & -0.103 & -0.0219 \\ & (-5.69) & (-3.33) & (-3.25) & (-0.78) & (-1.50) \\ \mbox{Time to Invest} & 0.110*** & 0.0966** & 0.0927*** & 0.00299 & -0.00496 \\ & (-6.88) & (7.93) & (7.99) & (0.07) & (-0.82) \\ \mbox{Country Characteristics} & & & & & & & & & & & & & & & & & & &$		(-0.75)	(-1.06)	(-1.06)	(0.72)	(-0.02)
Construction         (0.016)         (0.044)         (0.046)         (0.057)         (0.099)           Pre-EM Experience         -0.000147         -0.0000545         1.574           Time to Exit         -0.279***         -0.141***         -0.143**         -0.103         -0.0219           Time to Exit         -0.279***         -0.141***         -0.143**         -0.103         -0.0219           Time to Invest         0.110***         0.0996***         0.0997***         0.0029         -0.00406           Capital Inflow         -0.117         -0.217**         -0.240**         0.848**         0.03110           Nearby         -1.068         1.629*         1.714*         2.271         -0.324           Chitat         -0.349         0.986*         1.404*         0.110         -0.038           Distant         -0.349         0.986*         1.404*         0.110         -0.038           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           Iterarchy Distance         -1.378         -1.076*         -1.653*         2.502         0.399           Intellectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757      0	Co-investment	-0.111	-0.239	-0.255	0.676	0.0184
Pre-EM Experience         -0.000147         -0.000147         -0.000147         -0.000147           Time to Exit         -0.279***         -0.141***         -0.103         -0.0219           (5.69)         (5.33)         (3.25)         (0.78)         (1.50)           Time to Invest         0.110***         0.0996***         0.0927***         0.0029         -0.00496           Country Characteristics         (6.88)         (7.73)         (7.99)         (0.07)         (0.82)           Country Characteristics         (1.11)         (2.25)         (2.21)         (2.68)         (0.60)           Nearby         -1.068         1629*         1.714         2.271         -0.324           Capital Inflow         -0.117         -0.217**         -0.240**         0.848**         0.0311           Nearby         -1.068         1629*         1.714         2.271         -0.324           Cintal Millow         -0.1378         1.629*         1.714         2.271         -0.324           Bistant         -0.349         0.986*         1.140*         0.110         -0.033           Ibrearchy Distance         (1.21)         (2.11)         (2.14)         (0.48)         (0.16)           Ibrearchy Distance		(-0.16)	(-0.44)	(-0.46)	(0.45)	(0.09)
The number is the second sec	Pre-EM Experience	( 0.10)	-0.000147	-0.0000545	1.574	(0.07)
Time to Exit         -0.279***         -0.141***         -0.143**         -0.105         -0.0219           Time to Invest         0.110***         0.0906***         0.0927***         0.00299         -0.00496           Time to Invest         0.110***         0.0906***         0.0927***         0.00299         -0.00496           Capital Inflow         -0.117         -0.217**         -0.240**         0.848**         0.0311           Capital Inflow         -0.117         -0.217**         -0.240**         0.848**         0.0311           Nearby         -1.068         1.629*         1.714*         2.271         -0.324           (-1.04)         (2.35)         (2.41)         0.011         -0.034           Distant         -0.349         0.986*         1.140*         0.110         -0.038           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           Hierarchy Distance         -1.378         -1.076*         -1.653*         2.502         0.399           (-1.21)         (-2.11)         (-2.14)         (-2.14)         (0.48)         (-1.72)           Mastery Distance         -0.263         -0.932         -0.730         -2.061         0.757     <	The Envi Experience		(-0.83)	(-0.30)	(0.86)	
Interest Nate         (5.69)         (3.13)         (5.15)         (0.78)         (1.50)           Time to Invest         0.110***         0.0906***         0.0927***         0.00299         -0.00496           Country Characteristics         (6.88)         (7.93)         (7.99)         (0.07)         (6.82)           Capital Inflow         -0.117         -0.217**         -0.240**         0.848**         0.0311           Nearby         -1.068         1.629*         1.714*         2.271         -0.324           Nearby         -1.068         1.629*         1.714*         2.271         -0.324           Distant         -0.349         0.986*         1.140*         0.110         -0.0334           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           Itildechness distance         3.551         1.368         3.667         -1.005         -0.0901           Gorad         (0.73)         (1.21)         (-0.48)         (-0.16)         -0.105         -0.0901           Hielectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757           Distance         (-0.16)         (-1.14)         (-0.74)         (-0.79) <td>Time to Exit</td> <td>-0 279***</td> <td>-0 141***</td> <td>-0 143**</td> <td>-0.103</td> <td>-0.0219</td>	Time to Exit	-0 279***	-0 141***	-0 143**	-0.103	-0.0219
Time to Invest         0.10***         0.0906***         0.092***         0.0022***         0.0029         -0.00496           Country Characteristics		(-5.69)	(-3 33)	(-3.25)	(-0.78)	(-1.50)
Thick of the second s	Time to Invest	0 110***	0 0906***	0 0927***	0.00299	-0.00496
Country Characteristics         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.55)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.56)         (1.57)         (1.56)         (1.57)         (1.57)         (1.56)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.57)         (1.53)         (1.57)         (1.53)         (1.57)         (1.53)         (1.57)         (1.53)         (1.57)         (1.53)         (1.53)         (1.53)         (1.53)         (1.53)         (1.53)         (1.51)	Time to myest	(6.88)	(7.93)	(7.99)	(0.07)	(-0.82)
Owner of the construction of the constructi	Country Characteristics	(0.00)	(1.55)	(1.77)	(0.07)	( 0.02)
Community         Control         Contro         Control         Control         <	Capital Inflow	-0.117	-0 217**	-0 240**	0 848**	0.0311
Nearby         1.1068         1.6239*         (1.714*         (2.00)         (0.00)           Distant         -0.349         0.986*         1.140*         (0.11)         -0.034           Distant         -0.349         0.986*         1.140*         0.110         -0.0334           Label Control         (0.78)         (2.11)         (2.31)         (0.11)         (-0.08)           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           (0.78)         (0.73)         (1.21)         (-0.48)         (-0.16)         -0.091           Hierarchy Distance         -1.378         -1.076*         -1.653*         2.502         0.399           (-1.21)         (-2.11)         (-2.14)         (0.85)         (1.72)           Mastery Distance         -2.663         -0.935         -0.730         -2.061         0.757           Distance         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         -         -         -0.782***         0.174         0.173           First Time Investment         -0.652**         -0.782***         0.0171**         0.00901         0.00224           Qc	Suprai milow	(-1 11)	(-2 59)	(-2.81)	(2.68)	(0.60)
Interp       Interp<       Interp< <thinterp<< th=""> <thinterp<< th=""> <thint< td=""><td>Nearby</td><td>-1.068</td><td>1 629*</td><td>1 714*</td><td>2 271</td><td>-0.324</td></thint<></thinterp<<></thinterp<<>	Nearby	-1.068	1 629*	1 714*	2 271	-0.324
Distant         -0.349         0.986*         1.140*         0.110         -0.0334           Lindow         (0.58)         (2.11)         (2.31)         (0.11)         (-0.0334           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           (0.78)         (0.73)         (1.21)         (-0.48)         (-0.16)           Hierarchy Distance         -1.378 <b>-1.076* -1.653*</b> 2.502         0.399           (1.21)         (-2.11)         (-2.14)         (0.85)         (1.72)           Mastery Distance         -2.560         0.854         -0.957         -1.956         -0.838           (-0.56)         (0.48)         (-0.32)         (-0.34)         (-1.40)           Intellectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757           Distance         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         -         -0.782***         0.174         0.0209           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0224           (4.20)         (5.64)         (5.72)         (0.25	ivealby	(-1.04)	(2.35)	(2.41)	(0.71)	(-0.58)
Distant         100,000         11,100         100,000         100,000           Embeddedness distance         3,551         1,368         3,667         -1,005         -0,00901           Mierarchy Distance         -1,378         -1,076*         -1,653*         2,502         0,399           (1.21)         (-2,11)         (-2,14)         (0,85)         (1,72)           Mastery Distance         -2,560         0,854         -0,957         -1,956         -0,838           (-0.56)         (0,48)         (-0,32)         (-0,34)         (-1,46)           Intellectual Autonomy         -0,263         -0.935         -0.730         -2,061         0.757           Distance         -0,263         -0.935         -0.730         -2,061         0.757           Company Characteristics         -0.652**         -0.773***         -0.782***         0.174         0.173           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0229         (1,69)           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0024         (1,69)         (0,21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224	Distant	(-1.04)	0.986*	(2.+1) 1 140*	(0.71)	0.0334
Embeddedness distance         (0.53)         (2.11)         (2.11)         (0.11)         (0.0001)           Embeddedness distance         3.551         1.368         3.667         -1.005         -0.0901           (0.78)         (0.73)         (1.21)         (-0.48)         (-0.001)           Hierarchy Distance         -1.378         -1.076*         -1.653*         2.502         0.399           Mastery Distance         -2.560         0.854         -0.957         -1.956         -0.838           (-0.56)         (0.48)         (-0.32)         (-0.34)         (-1.40)           Intellectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757           0.510         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         -0.652**         -0.773***         -0.782***         0.174         0.173           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0229           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.6	Distant	-0.54)	(2 11)	(2.31)	(0.11)	-0.0334
Linecureducences uistance       5.551       1.506       5.007       -1.603       -0.0501         (0.78)       (0.73)       (1.21)       (-0.48)       (-0.16)         Hierarchy Distance       -1.578       -1.076*       -1.653*       2.502       0.399         (1.21)       (-2.14)       (0.85)       (1.72)         Mastery Distance       -2.560       0.854       -0.957       -1.956       -0.838         (-0.56)       (0.48)       (-0.32)       (-0.34)       (-1.46)         Intellectual Autonomy       -0.263       -0.935       -0.730       -2.061       0.757         Distance       (-0.16)       (-1.14)       (-0.74)       (-0.89)       (1.33)         Company Characteristics       -0.762***       -0.782***       0.174       0.173         First Time Investment       -0.652**       -0.773***       -0.782***       0.174       0.173         (2.299)       (-3.91)       (-3.90)       (0.20)       (1.69)       0.020         VC Dummy       0.0447       -0.0316       -0.0584       0.0269       0.021         Age       0.059***       0.0682***       0.0711***       0.00961       0.00224         (4.20)       (5.64) <td< td=""><td>Embaddadaass distance</td><td>(-0.58)</td><td>(2.11)</td><td>(2.31)</td><td>(0.11)</td><td>(-0.08)</td></td<>	Embaddadaass distance	(-0.58)	(2.11)	(2.31)	(0.11)	(-0.08)
Hierarchy Distance         -1.378         -1.076*         -1.653*         2.502         0.399           Mastery Distance         -1.210         (-2.11)         (-2.14)         (0.85)         (1.72)           Mastery Distance         -2.560         0.854         -0.957         -1.956         -0.838           (-0.56)         (0.48)         (-0.32)         (-0.34)         (-1.46)           Intellectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757           Distance         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.00226           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711**         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital	Embeddedness distance	(0.78)	(0.73)	(1.21)	-1.003	-0.0701
Interacting Distance       -1.376       -1.076*       -1.053*       2.302       0.399         Mastery Distance       (-1.21)       (-2.11)       (-2.14)       (0.85)       (1.72)         Mastery Distance       -2.560       0.848       -0.957       -1.956       -0.838         (-0.56)       (0.48)       (-0.32)       (-0.34)       (-1.40)         Intellectual Autonomy       -0.263       -0.935       -0.730       -2.061       0.757         Distance       (-0.16)       (-1.14)       (-0.74)       (-0.89)       (1.33)         Company Characteristics       (-0.16)       (-1.14)       (-0.74)       (0.49)       (1.53)         VC Dummy       0.0447       -0.0316       -0.0584       0.540       0.0269         (0.13)       (-0.13)       (-0.23)       (0.69)       (0.21)         Age       0.0595**       0.0682***       0.0711***       0.00961       0.00224         (4.20)       (5.64)       (5.72)       (0.25)       (0.53)         Invested Capital       -0.361*       -0.522***       -0.533***       -1.151**       -0.017*         (-2.39)       (-3.79)       (-3.73)       (-2.99)       (-0.40)       (-2.35)       0.117 <t< td=""><td>Uissessher Distance</td><td>(0.78)</td><td>(0.73)</td><td>(1.21)</td><td>(-0.46)</td><td>(-0.10)</td></t<>	Uissessher Distance	(0.78)	(0.73)	(1.21)	(-0.46)	(-0.10)
(-1.21)         (-2.11)         (-2.14)         (0.85)         (1.72)           Mastery Distance         -2.560         0.854         -0.957         -1.956         -0.838           (-0.56)         (0.48)         (-0.32)         (-0.34)         (-1.46)           Intellectual Autonomy         -0.263         -0.935         -0.730         -2.061         0.757           (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         (-2.99)         (-3.91)         (-3.90)         (0.20)         (1.69)           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0229           (0.13)         (-0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711**         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533**         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)         1.644         (2.30)         (2.42)         (1.10)         (0.89) <td>Hierarchy Distance</td> <td>-1.3/8</td> <td>-1.0/6*</td> <td>-1.055*</td> <td>2.502</td> <td>0.399</td>	Hierarchy Distance	-1.3/8	-1.0/6*	-1.055*	2.502	0.399
Mastery Distance       -2.500       0.854       -0.957       -1.956       -0.858         (-0.56)       (0.48)       (-0.32)       (-0.34)       (-1.46)         Intellectual Autonomy       -0.263       -0.935       -0.730       -2.061       0.757         Distance       (-0.16)       (-1.14)       (-0.74)       (-0.89)       (1.33)         Company Characteristics         First Time Investment       -0.652**       -0.773***       -0.782***       0.174       0.173         (-2.99)       (-3.91)       (-3.90)       (0.20)       (1.69)         VC Dummy       0.0447       -0.0316       -0.0584       0.0220         (age       0.0595***       0.0682***       0.0711***       0.00961       0.00224         Age       0.0595***       0.0682***       0.0711***       0.00961       0.00224         Age       0.0595***       0.0682***       0.572)       (0.25)       (0.53)         Invested Capital       -0.361*       -0.522***       -0.533***       -1.151**       -0.0197         (-2.39)       (-3.79)       (-3.73)       (-2.99)       (-0.40)         Deal Sequence       -0.00823***       0.000647*       0.000506       0.00284***	M · D'	(-1.21)	(-2.11)	(-2.14)	(0.85)	(1./2)
(-0.56)         (0.48)         (-0.32)         (-0.34)         (-1.40)           Intellectual Autonomy Distance         -0.263         -0.935         -0.730         -2.061         0.757           Company Characteristics         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         -0.652**         -0.773***         -0.782***         0.174         0.173           First Time Investment         -0.652**         -0.773***         -0.782***         0.174         0.173           (-2.99)         (-3.91)         (-3.90)         (0.20)         (1.69)           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0269           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***	Mastery Distance	-2.560	0.854	-0.957	-1.956	-0.858
Intellectual Autonomy Distance         -0.263         -0.935         -0.730         -2.061         0.757           Distance         (-0.16)         (-1.14)         (-0.74)         (-0.89)         (1.33)           Company Characteristics         -0.652**         -0.773***         -0.782***         0.174         0.173           First Time Investment         -0.652**         -0.773***         -0.782***         0.174         0.173           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0269           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Ye		(-0.56)	(0.48)	(-0.32)	(-0.34)	(-1.46)
(-0.16)       (-1.14)       (-0.74)       (-0.89)       (1.33)         Company Characteristics         First Time Investment       -0.652**       -0.773***       -0.782***       0.174       0.173         (-2.99)       (-3.91)       (-3.90)       (0.20)       (1.69)         VC Dummy       0.0447       -0.0316       -0.0584       0.540       0.0269         (0.13)       (-0.13)       (-0.23)       (0.69)       (0.21)         Age       0.0595***       0.0682***       0.0711***       0.00961       0.00224         Age       0.540       (5.64)       (5.72)       (0.25)       (0.53)         Invested Capital       -0.361*       -0.522***       -0.533***       -1.151**       -0.0197         (-2.39)       (-3.79)       (-3.73)       (-2.99)       (-0.40)         Deal Sequence       -0.00823***       0.000647*       0.000506       0.00284***       0.00117*         (Lub Deal Dummy       0.677       1.623*       1.809*       1.553       0.117         (0.84)       (2.30)       (2.42)       (1.10)       (0.89)         Intercept       Yes       Yes       Yes       Yes       Yes         Investment ye	Distance	-0.263	-0.935	-0.730	-2.061	0.757
Company Characteristics         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.09)         (4.01)         (4.09)         (4.09)	Distance	(-0.16)	(-1.14)	(-0.74)	(-0.89)	(1.33)
First Time Investment         -0.652**         -0.773***         -0.782***         0.174         0.173           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0269           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes	<b>Company Characteristics</b>					( )
(-2.99)         (-3.91)         (-3.90)         (0.20)         (1.69)           VC Dummy         0.0447         -0.0316         -0.0584         0.540         0.0269           (0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes         Yes         Yes           Industry FE         Yes         Yes         Yes         Yes         Yes <t< td=""><td>First Time Investment</td><td>-0.652**</td><td>-0.773***</td><td>-0.782***</td><td>0.174</td><td>0.173</td></t<>	First Time Investment	-0.652**	-0.773***	-0.782***	0.174	0.173
VC Dummy       0.0447       -0.0316       -0.0584       0.540       0.0269         (0.13)       (-0.13)       (-0.23)       (0.69)       (0.21)         Age       0.0595***       0.0682***       0.0711***       0.00961       0.00224         (4.20)       (5.64)       (5.72)       (0.25)       (0.53)         Invested Capital       -0.361*       -0.522***       -0.533***       -1.151**       -0.0197         (-2.39)       (-3.79)       (-3.73)       (-2.99)       (-0.40)         Deal Sequence       -0.00823***       0.000647*       0.000506       0.00284***       0.00117*         (-6.47)       (2.33)       (1.75)       (3.94)       (2.26)         Club Deal Dummy       0.677       1.623*       1.809*       1.553       0.117         (0.84)       (2.30)       (2.42)       (1.10)       (0.89)         Intercept       Yes       Yes       Yes       Yes       Yes       Yes       Yes         Investment year FE       Yes       Yes       Yes       Yes       Yes       Yes       Yes         Number of Countries       25       47       25       22       20         Number of Companics       9 268		(-2.99)	(-3.91)	(-3.90)	(0.20)	(1.69)
(0.13)         (-0.13)         (-0.23)         (0.69)         (0.21)           Age         0.0595***         0.0682***         0.0711***         0.00961         0.00224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           (-6.47)         (2.33)         (1.75)         (3.94)         (2.26)           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes         Yes         Yes           Number of Countries         25         47         25         22         20           Number of Observations         18 994         25 357         23 179         30 399         23 067 </td <td>VC Dummy</td> <td>0.0447</td> <td>-0.0316</td> <td>-0.0584</td> <td>0.540</td> <td>0.0269</td>	VC Dummy	0.0447	-0.0316	-0.0584	0.540	0.0269
Age         0.0595***         0.0682***         0.0711***         0.00961         0.0224           (4.20)         (5.64)         (5.72)         (0.25)         (0.53)           Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           (-6.47)         (2.33)         (1.75)         (3.94)         (2.26)           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes         Yes         Yes         Yes           Industry FE         Yes         Yes         Yes         Yes         Yes         Yes         Yes           Number of Companies         9268         9665         9243         422         601           Number of observations         18 994 <t< td=""><td>,</td><td>(0.13)</td><td>(-0.13)</td><td>(-0.23)</td><td>(0.69)</td><td>(0.21)</td></t<>	,	(0.13)	(-0.13)	(-0.23)	(0.69)	(0.21)
(4.20)       (5.64)       (5.72)       (0.25)       (0.53)         Invested Capital       -0.361*       -0.522***       -0.533***       -1.151**       -0.0197         (-2.39)       (-3.79)       (-3.73)       (-2.99)       (-0.40)         Deal Sequence       -0.00823***       0.000647*       0.000506       0.00284***       0.00117*         Club Deal Dummy       0.677       1.623*       1.809*       1.553       0.117         (0.84)       (2.30)       (2.42)       (1.10)       (0.89)         Intercept       Yes       Yes       Yes       Yes       Yes         Investment year FE       Yes       Yes       Yes       Yes       Yes         Number of Countries       25       47       25       22       20         Number of Observations       18 994       25 357       23 179       30 399       23 067	Age	0.0595***	0.0682***	0.0711***	0.00961	0.00224
Invested Capital         -0.361*         -0.522***         -0.533***         -1.151**         -0.0197           (-2.39)         (-3.79)         (-3.73)         (-2.99)         (-0.40)           Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes         Yes         Yes         Yes         Yes           Investment year FE         Yes         Yes         Yes         Yes         Yes           Number of Countries         25         47         25         22         20           Number of Observations         18 994         25 357         23 179         30 399         23 067	0	(4.20)	(5.64)	(5.72)	(0.25)	(0.53)
Involved explain       (-0.001       (-0.001       (-0.001       (-0.001)         Deal Sequence       (-0.00823***       0.000647*       0.000506       0.00284***       0.00117*         Club Deal Dummy       (-6.47)       (2.33)       (1.75)       (3.94)       (2.26)         Club Deal Dummy       0.677       1.623*       1.809*       1.553       0.117         (0.84)       (2.30)       (2.42)       (1.10)       (0.89)         Intercept       Yes       Yes       Yes       Yes         Investment year FE       Yes       Yes       Yes       Yes         Number of Countries       25       47       25       22       20         Number of Companies       9 268       9 665       9 243       422       601         Number of observations       18 994       25 357       23 179       30 399       23 067	Invested Capital	-0.361*	-0.522***	-0.533***	-1.151**	-0.0197
Deal Sequence         -0.00823***         0.000647*         0.000506         0.00284***         0.00117*           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes	invested Supra	(-2.39)	(-3.79)	(-3.73)	(-2.99)	(-0.40)
Club Deal Dummy         Constraint         Co	Deal Sequence	-0.00823***	0.000647*	0.000506	0.00284***	0.00117*
Club Deal Dummy         0.677         1.623*         1.809*         1.553         0.117           (0.84)         (2.30)         (2.42)         (1.10)         (0.89)           Intercept         Yes	D'em bequeinee	(-6.47)	(2.33)	(1.75)	(3.94)	(2.26)
Intercept         Yes         Y	Club Deal Dummy	0.677	1 623*	1 809*	1 553	0.117
InterceptYesYesYesYesYesInvestment year FEYesYesYesYesYesIndustry FEYesYesYesYesYesNumber of Countries2547252220Number of Companies9 2689 6659 243422601Number of observations18 99425 35723 17930 39923 067	Glub Dear Dunning	(0.84)	(2.30)	(2.42)	(1.10)	(0.89)
InterceptTesTesTesTesTesInvestment year FEYesYesYesYesYesIndustry FEYesYesYesYesYesNumber of Countries2547252220Number of Companies9 2689 6659 243422601Number of observations18 99425 35723 17930 39923 067	Intercent	Ves	Ves	Ves	Ves	Ves
Industry FEYesYesYesYesYesNumber of Countries2547252220Number of Companies9 2689 6659 243422601Number of observations18 99425 35723 17930 39923 067	Investment year FE	Ves	Ves	Ves	Ves	Ves
Number of Countries         25         47         25         22         20           Number of Companies         9 268         9 665         9 243         422         601           Number of observations         18 994         25 357         23 179         30 399         23 067	Industry FE	Ves	Vee	Vec	Vec	Vee
Number of Companies         9 268         9 665         9 243         422         601           Number of observations         18 994         25 357         23 179         30 399         23 067	Number of Countries	25	47	25	22	20
Number of observations         18 994         25 357         23 179         30 399         23 067	Number of Companies	9 268	9 665	9 243	422	601
	Number of observations	18 994	25 357	23 179	30 399	23 067

Figure 5: Re-arranged deal sequences sorted on first investments and vintage years of each fund within DMtoDM&EM subgroup of GPs. This figure shows the methodology for establishing new deal sequences to control for possible sequence gaps within reported deals. A fund is considered DM or EM based on the most observed frequency for either EM or DM within its portfolio companies' locations. Deals are sorted within each fund based on their investment dates. Funds are sorted for each GP in the DMtoDM&EM subgroup by vintage year.

	1	1 <sup>st</sup> Deal = EM Deal	2 <sup>nd</sup> Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 EM Fund 1
	2	1 <sup>st</sup> Deal = DM Deal	2 <sup>nd</sup> Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 EM Fund 2
	3	1 <sup>st</sup> Deal = DM Deal	$2^{nd}$ Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 EM Fund 3
	4	1 <sup>st</sup> Deal = EM Deal	$2^{nd}$ Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 DM Fund 1
New Sequence	5	1 <sup>st</sup> Deal = EM Deal	2 <sup>nd</sup> Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 DM Fund 2
Lagged Deal-level	6	1 <sup>st</sup> Deal = DM Deal	$2^{nd}$ Deal = EM Deal	3 <sup>rd</sup> Deal = DM Deal	 EM Fund 4
PMEs					
	i	1 <sup>st</sup> Deal = DM Deal	$2^{nd}$ Deal = EM Deal	$3^{rd}$ Deal = DM Deal	 EM Fund n
	j	1 <sup>st</sup> Deal = DM Deal	2 <sup>nd</sup> Deal = EM Deal	$3^{rd}$ Deal = DM Deal	 DM Fund m

Table 12 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME, accounting for previous EM deal performance on follow-on DM deal performance. This table shows the regression results for deal-level PMEs on lagged deal-level PMEs with the new sequence numbers (See figure 5). Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

	DMtoDM&EM						
	Given any DM/EM Deal Order	Given the first DM deal Followed by a DM deal After the last EM deal (EM-DM-DM sequence)	Given the first DM deal Followed by an EM deal After the last EM deal (EM-DM-EM sequence)				
Lead DM Deal PME	0.547***	0.545***	0.583***				
	(202.91)	(200.48)	(25.53)				
GP Characteristics		0.000	0.0704				
Size	-0.0864	-0.0931	-0.0786				
T 1 A (71'	(-1.60)	(-1./0)	(-0.50)				
Local Affiliation	-1.02/***	-1.032***	-0.9/1				
CD Direct Location and	(-5.12)	(-5.06)	(-1.52)				
GP Diffect Investment	-0.390	-0.413	0.936				
Cointroctmont	(-0.70)	(-0.72)	(0.83)				
Co-investment	-0.559	-0.539	0.300				
Dec EM Experience	(-0.03)	(-0.62)	(0.43)				
Pre-EM Experience	(0.21)	(0.63)	0.216				
Time to Exit	(0.21) 0 154***	0.05)	(0.10)				
Time to Exit	-0.134	-0.155	-0.0112				
Time to Invest	(-3.00) 0 0840***	0.0862***	-0.0256				
Time to invest	(7.00)	(7.06)	(-0.76)				
Country Characteristics	(1.00)	(7.00)	(0.70)				
Capital Inflow	-0 405***	-0 422***	0.299				
Capital Inflow	(-4.62)	(-4.72)	(1.23)				
Nearby	1.573*	1.685*	0.347				
	(2.20)	(2.30)	(0.15)				
Distant	0.454	0.604	-0.0594				
	(0.94)	(1.18)	(-0.08)				
Embeddedness distance	1.532	2.249	0.277				
	(0.82)	(0.73)	(0.19)				
Hierarchy Distance	-1.076*	-1.286	-0.0782				
	(-2.10)	(-1.63)	(-0.04)				
Mastery Distance	0.691	0.408	1.628				
	(0.39)	(0.14)	(0.41)				
Intellectual Autonomy Distance	-0.884	-0.990	-2.432				
	(-1.06)	(-0.98)	(-1.52)				
Company Characteristics							
First Time Investment	-0.643**	-0.636**	-0.815				
	(-3.10)	(-3.03)	(-1.22)				
VC Dummy	0.0783	0.0878	-0.338				
	(0.30)	(0.33)	(-0.56)				
Age	0.0604***	0.0631***	0.0138				
	(4.98)	(5.04)	(0.51)				
Invested Capital	-0.359**	-0.383**	-0.462				
	(-2.59)	(-2.66)	(-1.67)				
Club Deal Dummy	0.544	0.745	-0.469				
-	(0.77)	(0.99)	(-0.48)				
Intercept	Yes	Yes	Yes				
Investment year FE	Yes	Yes	Yes				
Industry FE	Yes	Yes	Yes				
Number of Countries	47	25	22				
Number of chapters	9 663 E0 104	y 234 E1 100	429				
INUMBER OF ODSERVATIONS	52 184	51 190	994				

or the product of the state of	DMto	DM	DMtoDM&EM					EMto	EM	
			Ove	rall	DM	toDM	DMto	DEM		
GP Characteristics	Pre-2000	Post-2000	Pre-2000	Post-2000	Pre-2000	Post-2000	Pre-2000	Post-2000	Pre-2000	Post-2000
Size	0.524*	-0.0333	-0.0869	0.0295	-0.0908	0.0181	-0.00808	0.156	-0.000334	-0.123
	(2.17)	(-0.65)	(-0.48)	(0.84)	(-0.49)	(0.51)	(-0.18)	(0.64)	(-1.63)	(-1.83)
Local Affiliation	-2.844**	-0.696***	-1.878**	-0.349*	-1.888**	-0.389*	-2.518***	-0.187	-0.122***	-0.907***
	(-3.22)	(-4.84)	(-3.22)	(-2.30)	(-3.23)	(-2.54)	(-9.06)	(-0.18)	(-49.51)	(-4.30)
GP Direct Investment		0.0144		-0.0115		-0.0501		1.477		0.306
		(0.04)		(-0.04)		(-0.17)		(0.89)		(0.74)
Co-investment		-0.180		-0.244		-0.287		1.329		0.548
		(-0.57)		(-0.83)		(-0.96)		(0.79)		(1.37)
Pre-EM Experience			0.000855	-0.000989***	0.000903	-0.000914***	-0.265	3.088		
			(1.74)	(-6.51)	(1.82)	(-5.77)	(-0.24)	(0.77)		
Time to Exit	-0.697***	-0.335***	0.0425	-0.373***	0.0407	-0.394***	0.0751	-0.238	-0.00908	-0.0396
	(-4.30)	(-7.72)	(0.31)	(-7.92)	(0.29)	(-8.06)	(0.43)	(-1.33)	(-0.20)	(-0.97)
Time to Invest	0.458***	0.0404***	0.334***	0.0286***	0.335***	0.0288***	0.0649***	-0.00111	0.0000584	-0.0281*
	(6.58)	(4.97)	(7.44)	(3.84)	(7.43)	(3.82)	(3.38)	(-0.02)	(0.52)	(-2.45)
Country Characteristics										
Capital Inflow	0.666*	-0.199**	-0.701**	0.0997	-0.709**	0.0611	0.00777	0.721	0.0693***	0.275*
	(2.17)	(-2.77)	(-2.86)	(1.44)	(-2.87)	(0.86)	(0.09)	(1.94)	(33.16)	(2.47)
Nearby	5.353	-0.254	10.80***	2.493***	10.98***	2.444***	-3.090	2.388	-0.998	-1.948
	(0.71)	(-0.43)	(3.35)	(5.19)	(3.37)	(5.05)	(-0.56)	(0.63)	(-0.82)	(-1.46)
Distant	2.991	-0.447	7.452***	0.392	7.661***	0.401	-2.211***	-0.250		0.0360
	(1.24)	(-1.24)	(3.91)	(1.22)	(3.96)	(1.19)	(-6.88)	(-0.21)		(0.03)
Embeddedness distance	-5.427	-0.456	3.068	-0.213	9.728	1.222	-1.193	-0.496	0.160	0.304
	(-0.13)	(-0.19)	(0.24)	(-0.13)	(0.52)	(0.56)	(-0.36)	(-0.19)	(0.05)	(0.26)
Hierarchy Distance	-1.844	0.0224	-5.141	-0.411	-6.838	-0.711	-4.703	3.887	1.604	1.264**
	(-0.19)	(0.04)	(-1.64)	(-0.97)	(-1.54)	(-1.27)	(-0.87)	(1.05)	(0.65)	(2.69)
Mastery Distance	4.253	-0.242	9.725	2.383	4.964	0.941	9.615	-3.794	-2.739	-0.652
	(0.10)	(-0.10)	(0.75)	(1.54)	(0.27)	(0.44)	(0.96)	(-0.53)	(-0.43)	(-0.58)
Intellectual Autonomy Distance	-3.757	-1.003	-7.753	-1.617**	-7.889	-1.184	0.625	-2.894	1.119	0.821
	(-0.26)	(-1.10)	(-1.48)	(-2.63)	(-1.29)	(-1.71)	(0.20)	(-0.99)	(0.26)	(0.68)
Company Characteristics										
First Time Investment	-1.113	-0.892***	-2.176***	-0.154	-2.174***	-0.146	-0.753***	-0.129	-0.00140	0.289
	(-1.76)	(-6.18)	(-4.02)	(-0.87)	(-4.00)	(-0.81)	(-4.85)	(-0.12)	(-1.61)	(1.44)
VC Dummy	1.020	-0.0400	-1.567	0.0649	-1.621	0.0685	-0.741***	0.0104	1.752***	0.0114
	(0.88)	(-0.19)	(-1.80)	(0.37)	(-1.85)	(0.38)	(-3.64)	(0.01)	(102.74)	(0.04)
Age	0.0693	0.0535***	0.0299	0.0570***	0.0321	0.0603***	0.0442	-0.0306	-0.00195**	0.0124
	(1.16)	(5.11)	(0.66)	(4.71)	(0.70)	(4.84)	(0.78)	(-0.60)	(-2.75)	(1.08)
Invested Capital	-1.300*	0.00896	-1.371**	-0.497***	-1.441**	-0.448**	-0.185	-1.031*	0.0937	-0.186
	(-2.43)	(0.07)	(-2.92)	(-3.54)	(-3.03)	(-3.05)	(-0.38)	(-2.15)	(0.76)	(-1.61)
Deal Sequence	-0.0426***	0.0176***	-0.00181*	0.00337***	-0.00187*	0.00327***	0.00197***	0.00409***	0.000958***	0.00625***
	(-11.65)	(19.96)	(-2.34)	(13.86)	(-2.40)	(12.80)	(14.37)	(4.43)	(71.90)	(5.16)
Club Deal Dummy	2.263	1.173	-0.0716	1.567*	0.0312	1.629*	-1.705	1.949	-0.154	0.664*
	(0.75)	(1.84)	(-0.03)	(2.16)	(0.01)	(2.08)	(-0.81)	(1.10)	(-0.27)	(2.01)
Intercept	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Investment year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Number of Countries	18	26	35	45	21	25	14	20	12	20
Number of Companies	4 205	6 640	4 500	6 770	4 440	6 355	60	415	36	671
Number of observations	18 994	25 357	23 179	30 399	23 067	29 456	112	943	84	1 269

Table 13 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME, differentiated by investment period. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

Table 14 : Multilevel Linear Model Estimates for GP, Country and Portfolio Company effects on Deal-level PME, considering change in investment style of GP when investing in EM. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

	DMtoDM&EM - DMtoEM Subsample estimates							
	Deal PME	Multiple of Invested Capital	Exit Rate					
Style Shift	-0.905	-0.307	0.00327					
	(-1.23)	(-1.32)	(0.19)					
GP Characteristics								
Size	-0.0227	-0.122***	0.00406***					
	(-0.65)	(-17.58)	(11.89)					
Local Affiliation	-0.0795**	-0.129***	-0.00323***					
	(-2.74)	(-24.18)	(-11.77)					
GP Direct Investment	0.256	0.121	0.0254***					
	(1.12)	(1.70)	(7.21)					
Co-investment	-0.139	0.0197	0.0278***					
	(-0.65)	(0.27)	(7.72)					
Pre-EM Experience	1.260***	0.430***	0.0220***					
1	(13.03)	(30.60)	(31.17)					
Time to Exit	-0.261	0.0413	0.00947***					
	(-1.66)	(1.64)	(6.26)					
Time to Invest	0.0822	0.000372	0.00128*					
	(1.68)	(0.04)	(2 57)					
Country Characteristics	(1.00)		(2.07)					
Capital Inflow	0.145*	-0.0675***	0.0218***					
onprim millo "	(2.51)	(-6.99)	(44 28)					
Nearby	0.102	0.183	-0.0375***					
ivenby	(0.35)	(1.88)	(-7.36)					
Distont	2 310***	0 336***	0.0628***					
Distant	-2.519	(5.37)	(10.42)					
Embaddadaass distance	(-0.77)	(5.57)	0.071/**					
	(0.80)	(2.02)	(2.20)					
History and an Distance	(0.80)	(-2.92)	(3.20)					
Hierarchy Distance	-0.211	-0.232**	-0.03/8****					
M & D'	(-0.64)	(-2.53)	(-0.53)					
Mastery Distance	0.124	0.860*	-0.0485*					
	(0.13)	(2.41)	(-2.12)					
Intellectual Autonomy Distance	-0.594	-1.459***	-0.0422***					
	(-1.44)	(-10.82)	(-5.48)					
Company Characteristics								
First Time Investment	-2.460***	-0.630***	-0.115***					
	(-4.90)	(-8.25)	(-29.65)					
VC Dummy	-0.0317	0.623***	0.127***					
	(-0.25)	(19.48)	(79.18)					
Age	0.00110*	-0.000561***	-0.000106***					
	(2.17)	(-5.03)	(-16.58)					
Invested Capital	0.000505	0.0000597	-0.0000518***					
	(1.44)	(0.73)	(-10.53)					
Deal Sequence	-1.978***	-0.954***	-0.0910***					
	(-4.31)	(-14.30)	(-26.73)					
Club Deal Dummy	-0.0348	0.0169	0.000439					
	(-0.42)	(1.39)	(0.58)					
Intercept	Yes	Yes	Yes					
Investment year FE	Yes	Yes	Yes					
Industry FE	Yes	Yes	Yes					
Number of Countries	22	26	26					
Number of Companies	448	6 640	9 677					
Number of observations	3 967	53 042	55 306					

**Table 15 : Robustness of results using other performance measures.** This table shows the multilevel linear model estimates for GP, Country and Portfolio Company effects on exit rates and multiples of invested capital as performance measures. Variable definitions are given in Appendix 1. T-statistics are given between brackets. One, two and three asterisks denote a 10%, 5% and 1% significance level respectively.

	Exit Rate					Multiple of invested Capital				
-	DMtoDM		DMtoDM&EM		EMtoEM	DMtoDM DMtoDM&EM				EMtoEM
		Overall	DMtoDM	DMtoEM		_	Overall	DMtoDM	DMtoEM	
GP Characteristics										
Size	0.00000695	-0.0000442**	-0.0000412**	-0.0000678	-0.00000527	-2.81e-09***	-2.48e-10	-0.0952***	5.54e-09**	1.12e-08***
	(0.49)	(-2.83)	(-2.59)	(-1.55)	(-0.27)	(-6.43)	(-0.78)	(-13.82)	(2.93)	(5.24)
Local Affiliation	-0.0000360	0.0000750	0.0000739	0.0000980	-0.000239***	1.40e-08***	6.63e-09***	0.177***	2.36e-08**	-2.13e-08**
	(-0.80)	(1.38)	(1.34)	(0.51)	(-3.71)	(10.40)	(5.76)	(7.05)	(3.15)	(-3.16)
GP Direct Investment	-0.0000178	0.000187	0.000204	-0.0000418	0.00000930	-1.70e-09	4.31e-09	0.118	-5.80e-10	1.53e-08
	(-0.14)	(1.10)	(1.16)	(-0.13)	(0.07)	(-0.42)	(1.29)	(1.61)	(-0.04)	(1.09)
Co-investment	0.0000536	0.000269	0.000286	-0.000124	0.0000616	-4.54e-09	1.47e-09	0.00614	4.85e-09	1.61e-08
	(0.43)	(1.54)	(1.59)	(-0.39)	(0.50)	(-1.13)	(0.43)	(0.08)	(0.34)	(1.19)
Pre-EM Experiance		-2.55e-08	-2.56e-08	-0.000275	. ,		-7.35e-12***	0.0000754***	-2.91e-09	
*		(-0.55)	(-0.54)	(-0.52)			(-7.37)	(3.38)	(-0.18)	
Time to Exit	0.0000256	0.00000793	0.00000760	-0.00000488	0.00000408	4.28e-11	-4.30e-10*	0.00293	1.09e-09	7.10e-10
	(1.67)	(1.11)	(1.05)	(-0.12)	(0.31)	(0.18)	(-2.17)	(0.64)	(1.06)	(0.67)
Time to Invest	-0.00000376	3.77e-08	0.000000296	-0.00000313	0.000000268	7.84e-11	6.83e-10***	0.00541***	4.54e-10	-4.81e-10
	(-1.33)	(0.01)	(0.08)	(-0.34)	(0.08)	(0.89)	(9.62)	(3.51)	(1.10)	(-1.25)
Country Characteristics	(	(010-)	(0100)	( 0.0.1)	(0.00)	(0101)	(,,,,_)	(0.0.1)	()	()
Capital Inflow	0.0000281	0.0000153	0.0000183	-0.0000518	0.0000414	1 77e-09**	7 44e-10	-0.0670***	1.99e-09	-9 55e-09**
Supital Innow	(1.46)	(0.65)	(0.76)	(-0.78)	(1 27)	(3.14)	(1.50)	(-6.16)	(0.72)	(-2.89)
Noorby	0.000122	0.00127***	0.00107***	0.00376***	0.0000465	0.050.00	1 340 08**	0.436***	5.000.08	(=2.07) 8 880 08
Inearby	(0.40)	(F. OF)	0.0010/****	(4.99)	0.0000403	-9.05e-09	1.546-00***	(4.20)	5.000-08	0.000-00
Distant	0.000502***	0.0000266	(4.91)	(4.00)	0.0000284	(-1.03)	2 22- 00	(4.39)	(1.75)	(1.23)
Distant	-0.000595***	-0.0000266	0.0000242	-0.0000858	0.0000284	1.26e-08***	-3.32e-09	0.422***	1.21e-08	5.15e-08
	(-4.46)	(-0.20)	(0.17)	(-0.37)	(0.07)	(4.21)	(-1.17)	(6.42)	(1.46)	(0.48)
Embeddedness distance	0.0000618	0.00269***	0.00529***	-0.00117/	0.0000246	1.01e-08	-4.30e-08**	-0.443	1./5e-08	0.000000144***
	(0.06)	(3.58)	(5.63)	(-1.03)	(0.07)	(0.44)	(-3.09)	(-0.94)	(1.06)	(3.90)
Hierarachy Distance	0.0000294	-0.000547**	-0.00115***	0.000862	0.0000753	1.01e-08	2.51e-08***	-0.594***	-2.86e-08	-0.000000106***
	(0.12)	(-2.84)	(-4.85)	(0.46)	(0.51)	(1.78)	(6.94)	(-5.05)	(-1.24)	(-6.74)
Mastery Distance	-0.00000390	-0.00463***	-0.00727***	0.00138	-0.000115	-1.90e-08	-1.86e-08	0.654	5.06e-09	5.36e-08
	(-0.00)	(-6.18)	(-7.89)	(0.44)	(-0.33)	(-0.83)	(-1.34)	(1.41)	(0.11)	(1.47)
Intellectual Autonomy Distance	0.000290	0.00227***	0.00291***	-0.00200	0.000147	2.30e-08**	2.01e-09	-1.731***	1.06e-08	-0.000000103**
	(0.79)	(7.79)	(9.28)	(-1.34)	(0.40)	(2.78)	(0.35)	(-11.53)	(0.58)	(-2.67)
Company Characteristics				· · · ·	. ,					
First Time Investment	0.0000314	-0.0000516	-0.0000567	0.000133	-0.0000728	-7.75e-09***	-5.16e-09***	0.249***	3.62e-09	-5.69e-09
	(0.82)	(-0.86)	(-0.93)	(0.71)	(-1.22)	(-6.52)	(-4.27)	(9.48)	(0.46)	(-0.92)
VC Dummy	-0.0000566	0.0000223	0.0000545	-0.000280	0.0000575	-6.70e-09***	7.86e-09***	0.837***	-5.77e-09	-4.31e-10
	(-0.89)	(0.30)	(0.72)	(-1.60)	(0.70)	(-3.66)	(5.32)	(26.09)	(-0.82)	(-0.05)
Age	-0.00000131	1 18e-08	0.000000458	-0.00000608	-0.000000148	-7 90e-11	-8 76e-11	-0.00745***	4 29e-11	4 08e-10
1.80	(-0.32)	(0.00)	(0.19)	(-0.48)	(-0.04)	(-1.13)	(-1.46)	(-5.47)	(0.14)	(1.29)
Invested Capital	0.0000290	0.0000264	0.0000266	0.0000596	0.0000170	7 38e 10	1 130 00	0.0022***	5 30e 00	2 320 10
nivested Capitai	(0.67)	-0.0000204	-0.0000200	-0.0000570	-0.000170	(1.00)	(1.64)	(5.85)	(1.74)	-2.520-10
Deal Section as	0.00000120	0.5%, 00	(-0.93) 5 00a 00	0.000000147	(-0.40)	(1.00) 2.04a 11***	(-1.04)	0.0000452	(-1.74)	(-0.07)
Deal Sequence	-0.00000120	-9.366-09	-5.008-09	-0.00000147	-4.396-06	5.040-11444	-2./30-12	-0.0000432	-1.116-11	-/.516-11
	(-0.51)	(-0.14)	(-0.07)	(-0.85)	(-0.11)	(4.01)	(-1./0)	(-1.29)	(-1./3)	(-1.93)
Club Deal Dummy	0.000000321	-0.000196	-0.0000568	-0.00117**	-0.0000683	-3.94e-09	3.3/e-09	0.338***	5.30e-09	1.46e-08
	(0.00)	(-1.37)	(-0.38)	(-2./1)	(-0.63)	(-1.00)	(0.97)	(4.17)	(0.47)	(1.58)
Intercept	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Investment year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Number of Countries	26	47	25	22	21	26	47	25	22	20
Number of Companies	9 987	10 213	9 746	467	739	9 775	9 987	9 535	452	692
Number of observations	46 068	56 437	55 345	1 092	1 439	44 823	54 124	53 064	1 060	1 364